Data distributions in magnetic resonance images: a review

A.J. den Dekker^{a,b,*}, J. Sijbers^a

^aiMinds-Vision Lab, University of Antwerp, Universiteitsplein 1, N.1, 2610 Wilrijk, Belgium ^bDelft Center for Systems and Control, Delft University of Technology, Mekelweg 2, 2628 CD Delft, The Netherlands

Abstract

Many image processing methods applied to magnetic resonance (MR) images directly or indirectly rely on prior knowledge of the statistical data distribution that characterizes the MR data. Also, data distributions are key in many parameter estimation problems and strongly relate to the accuracy and precision with which parameters can be estimated. This review paper provides an overview of the various distributions that occur when dealing with MR data, considering both single coil and multiple coil acquisition systems. The paper also summarizes how knowledge of the MR data distributions can be used to construct optimal parameter estimators and answers the question as to what precision may be achieved ultimately from a particular MR image. *Keywords:* magnetic resonance imaging, statistical data distributions, parameter estimation, noise modeling

Preprint submitted to Physica Medica: European Journal of Medical Physics June 8, 2014

^{*}Corresponding author. Tel. +32 (0) 3 265 2869 E-mail address: Arjan.denDekker@uantwerpen.be (A.J. den Dekker)

1. Introduction

Magnetic Resonance Imaging (MRI) is the diagnostic tool of choice in biomedicine. It is able to produce high-quality three-dimensional images containing an abundance of physiological, anatomical and functional information. A voxel's grey level within an MR image represents the amplitude of the radio frequency signal coming from the hydrogen nuclei (protons) within that voxel. To draw reliable diagnostic conclusions from MR images, visual inspection alone is often insufficient. Quantitative data analysis is required to extract the information needed. Such an analysis can almost without exception be formulated as a parameter estimation problem. The parameters of interest can simply be the values of the true MR signal underlying the noise corrupted data points [1-3], but also proton densities (in the construction of proton density maps [4, 5], relaxation time constants (in the construction of T_1, T_2 and T_2^* maps [4–11]) or diffusion parameters (in Diffusion MRI) [12– 14]. Different estimators can be constructed to estimate one and the same parameter, but it is well known that the best estimators (in terms of accuracy and precision) are constructed by properly taking the statistical distribution of the data into account. Hence, knowledge of the MRI data distribution is of vital importance.

This review paper gives an overview of the various distributions that occur when dealing with MR data, considering both single coil and multiple coil systems. The paper also summarizes how knowledge of these distributions can be used to construct optimal estimators and to answer the question as to what precision may be achieved ultimately from a particular MR image.

The organization of the paper is as follows. Section 2 briefly reviews MR

signal detection and introduces a statistical model of the complex valued raw MR data acquired in the so-called k-space (i.e., the spatial frequency domain). Section 3 then describes the statistical distribution of the reconstructed images in the spatial domain, assuming the data have been acquired using a single-coil system. Complex images as well as magnitude and phase images, which can be constructed from the complex images straightforwardly, are considered. Since image acquisition with multiple coils is becoming more and more common nowadays, section 4 describes the distribution of complex and magnitude images acquired with multiple coil-systems. Section 5 reviews the theory that explains how knowledge of the distribution of the MR images can be used to i) derive a lower bound on the variance of any unbiased estimator of parameters from these images (the so-called Cramér-Rao Lower Bound), and (ii) to construct the maximum likelihood (ML) estimator, which attains this lower bound at least asymptotically. In section 6, this theory is applied to i) derive the CRLB for unbiased estimation of the underlying true signal amplitude from (single coil and multiple-coil) magnitude images and, (ii) derive the ML estimator for this estimation problem. In section 7, conclusions are drawn.

Notation: Throughout this paper, vectors will be underlined and matrices will be expressed in capital letters. Furthermore, random variables (RVs) will be expressed in bold face. The operators $\mathbb{E}[\cdot]$ and $Var(\cdot)$ denote the expectation and variance of a random variable, respectively. The real part of a complex valued variable z is denoted as z_R and the imaginary part as z_I . The complex conjugate of X is denoted as X^* and the transpose and complex conjugate transpose of X are denoted as X^T and X^H , respectively.

Furthermore, we use $f_{\mathbf{x}}(x)$ to denote the probability density function (PDF) of the random variable \mathbf{x} . The conditional PDF of the RV \mathbf{x} conditioned on the RV \mathbf{y} is denoted as $f_{\mathbf{x}|\mathbf{y}}(x|y)$. The modified Bessel function of the first kind of order ν is denoted as $I_{\nu}(\cdot)$. The symbol *i* denotes $\sqrt{-1}$.

2. Signal detection and modeling

This section briefly reviews the mathematics behind signal detection in MRI and describes the concepts of signal demodulation and quadrature detection. The section is to a large extent based on [15–19]. For a more comprehensive description, the reader is referred to those references. The final purpose of the section is to introduce a statistical model of the detected MR signal.

2.1. Modeling the noise free signal

In MRI, an object is placed in a strong static, external, homogeneous magnetic field \underline{B}_0 that polarizes the protons in the object, yielding a net magnetic moment oriented parallel to \underline{B}_0 . Let's assume that \underline{B}_0 points in the z-direction. Next, a radio frequency pulse is applied that generates another, oscillating magnetic field \underline{B}_1 perpendicular to \underline{B}_0 . This so-called excitation field tips away the net magnetic moment from the z-axis, producing a magnetization component transverse to the static field. This transverse magnetization component precesses at the so-called Larmor frequency

$$\omega_0 = \gamma |\underline{B}_0|,$$

with γ the gyromagnetic ratio. This precessing magnetization vector induces a voltage in the receiver/detector coil (a conducting loop). Spatial information can be encoded in the received signal by augmenting \underline{B}_0 with additional, spatially varying magnetic fields. These so-called gradient fields vary linearly in space and are denoted as G_x , G_y and G_z . For example, when G_x is applied, the strength of the static magnetic field will vary with position in the x-direction as $|\underline{B}_z(x)| = |\underline{B}_0| + G_x x$, where the subscript z is used to denote that the magnetic field points in the z-direction. In this way, gradient fields can be used to to make the precession frequency vary linearly in space. MRI signal detection is based on Faraday's law of electromagnetic induction and the principle of reciprocity [15]. Assuming a static inhomogeneous magnetic field pointing in the z-direction, the (noise free) voltage signal v(t) in the receiver coil is related to the transverse magnetization distribution $\underline{M}_{xy}(\underline{r}, t)$ of the object by the expression [15]

$$v(t) = \int_{object} \omega(\underline{r}) |\underline{B}_{r,xy}(\underline{r})| |\underline{M}_{xy}(\underline{r}, 0_{+})| e^{-t/T_{2}(\underline{r})}$$

$$\cos\left[-\omega(\underline{r})t + \phi_{e}(\underline{r}) - \phi_{r}(\underline{r}) + \frac{\pi}{2}\right] d\underline{r}$$
(1)

with $\underline{r} = (x, y, z)^T$ the position in the laboratory frame, $t = 0_+$ the time instant immediately after the excitation pulse, $\omega(\underline{r})$ the free precession frequency, T_2 a relaxation time constant, $\underline{B}_{r,xy}(\underline{r})$ the detection sensitivity of the coil, $\phi_r(\underline{r})$ the reception phase angle, and $\phi_e(\underline{r})$ the initial phase shift introduced by RF excitation. The detection sensitivity $\underline{B}_{r,xy}(\underline{r})$ is defined as the xy vector component of the field generated at \underline{r} by a unit current in the coil. The phase contributions $\phi_r(\underline{r})$ and $\phi_e(\underline{r})$ take a value between 0 and 2π depending on the direction of, respectively, $\underline{B}_{r,xy}(\underline{r})$ and $\underline{M}_{xy}(\underline{r}, 0_+)$ in the transverse plane [15]. Assuming that a frequency encoding gradient G_x was turned on during the signal read out (i.e., during data acquisition), we have

$$\omega(\underline{r}) = \omega_0 + \Delta \omega(\underline{r}), \qquad (2)$$

with

$$\Delta\omega(\underline{r}) = \gamma G_x x,\tag{3}$$

where $\Delta\omega(\underline{r})$ is the spatially varying resonance frequency in the Larmorrotating frame, i.e., the coordinate system whose transverse plane is rotating clockwise at an angular frequency ω_0 [15]. Furthermore, if we assume that a so-called phase encoding gradient G_y was turned on for a time interval T_{pe} before the signal read out, we have to add a position dependent initial phase contribution $\phi_{pe}(\underline{r})$ to v(t):

$$v(t) = \int_{object} \omega(\underline{r}) |\underline{B}_{r,xy}(\underline{r})| |\underline{M}_{xy}(\underline{r}, 0_{+})| e^{-t/T_{2}(\underline{r})}$$

$$\cos\left[-\omega(\underline{r})t - \phi_{pe}(\underline{r}) + \phi_{e}(\underline{r}) - \phi_{r}(\underline{r}) + \frac{\pi}{2}\right] d\underline{r}, \qquad (4)$$

with

$$\phi_{pe}(\underline{r}) = \gamma G_y y T_{pe}.$$
(5)

MR image reconstruction concerns the inverse problem of reconstructing the transverse magnetization distribution $\underline{M}_{xy}(\underline{r},t)$ from the voltage signal v(t). If we assume that a slice selective gradient G_z has been applied in the z-direction during the excitation period, only protons in the selected slice (at, say, $z = z_0$) are excited, so that $\underline{M}_{xy}(x, y, z_0, t) = \underline{M}_{xy}(x, y, t)$ [18]. The MRI reconstruction problem then reduces to producing a spatial map in two dimensions. Assuming that $|\underline{M}_{xy}(\underline{r}, 0_+)|e^{-t/T_2(\underline{r})}$ is relatively constant during data acquisition, Equation (4) can be simplified to

$$v(t) = \int_{object} \omega(\underline{r}) |\underline{B}_{r,xy}(\underline{r})| |\underline{M}_{xy}(\underline{r}, t_{acq})| \cos\left[-\omega(\underline{r})t - \phi_{pe}(\underline{r}) + \phi_{e}(\underline{r}) - \phi_{r}(\underline{r}) + \frac{\pi}{2}\right] d\underline{r}$$
(6)

with t_{acq} the time at the center of the acquisition and

$$\underline{M}_{xy}(\underline{r}, t_{acq}) = |\underline{M}_{xy}(\underline{r}, 0_+)| e^{-t_{acq}/T_2(\underline{r})} e^{i\phi_e(\underline{r})}.$$
(7)

In practice, $\Delta\omega(\underline{r}) \ll \omega_0$ and v(t) is a high frequency bandpass signal centered about the frequency $\pm\omega_0$. The high-frequency nature of v(t) may cause unnecessary problems for electronic circuits in later processing stages [15]. In practice, these problems are circumvented by exploiting the following property of the bandpass signal v(t). It can be shown that the bandpass signal v(t) can be represented as [19]

$$v(t) = \Re \left[\tilde{v}(t) \exp(i\omega_0 t) \right], \tag{8}$$

where $\Re[z]$ denotes the real part of the complex number z and $\tilde{v}(t)$ is the so-called complex envelope of v(t), which can be written as

$$\tilde{v}(t) = \tilde{v}_R(t) + i\tilde{v}_I(t), \tag{9}$$

with

$$\tilde{v}_{R}(t) = \int_{object} \omega(\underline{r}) |\underline{B}_{r,xy}(\underline{r})| |\underline{M}_{xy}(\underline{r}, t_{acq})| \\ \cos\left[-\Delta\omega(\underline{r})t - \phi_{pe}(\underline{r}) + \phi_{e}(\underline{r}) - \phi_{r}(\underline{r}) + \frac{\pi}{2}\right] d\underline{r}, \quad (10)$$

and

$$\tilde{v}_{I}(t) = \int_{object} \omega(\underline{r}) |\underline{B}_{r,xy}(\underline{r})| |\underline{M}_{xy}(\underline{r}, t_{acq})| \\ \sin\left[-\Delta\omega(\underline{r})t - \phi_{pe}(\underline{r}) + \phi_{e}(\underline{r}) - \phi_{r}(\underline{r}) + \frac{\pi}{2} \right] d\underline{r}.$$
(11)

The signal $\tilde{v}_R(t)$ is called the *in-phase* component and $\tilde{v}_I(t)$ is called the *quadrature* component of v(t). Note that it follows from (8) and (9) that the original bandpass signal v(t) can be written in terms of $\tilde{v}_R(t)$ and $\tilde{v}_I(t)$ as:

$$v(t) = \tilde{v}_R(t)\cos(\omega_0 t) - \tilde{v}_I(t)\sin(\omega_0 t).$$
(12)

Since in practice $\Delta \omega(\underline{r}) \ll \omega_0$, Eqs. (10) and (11) can be simplified to

$$\tilde{v}_{R}(t) = \omega_{0} \int_{object} |\underline{B}_{r,xy}(\underline{r})| |\underline{M}_{xy}(\underline{r}, t_{acq})| \\ \cos\left[-\Delta\omega(\underline{r})t - \phi_{pe}(\underline{r}) + \phi_{e}(\underline{r}) - \phi_{r}(\underline{r}) + \frac{\pi}{2}\right] d\underline{r}, \quad (13)$$

and

$$\tilde{v}_{I}(t) = \omega_{0} \int_{object} |\underline{B}_{r,xy}(\underline{r})| |\underline{M}_{xy}(\underline{r}, t_{acq})| \\ \sin\left[-\Delta\omega(\underline{r})t - \phi_{pe}(\underline{r}) + \phi_{e}(\underline{r}) - \phi_{r}(\underline{r}) + \frac{\pi}{2}\right] d\underline{r}, \qquad (14)$$

and the complex envelope can be written as

$$\tilde{v}(t) = \omega_0 \int_{object} |\underline{B}_{r,xy}(\underline{r})| |\underline{M}_{xy}(\underline{r}, t_{acq})| e^{-i\left[\Delta\omega(\underline{r})t + \phi_{pe}(\underline{r}) - \phi_e(\underline{r}) + \phi_r(\underline{r}) - \frac{\pi}{2}\right]} d\underline{r}.$$
 (15)

Note that both $\tilde{v}_R(t)$ and $\tilde{v}_I(t)$ are low-pass signals. In practice, the signals $\tilde{v}_R(t)$ and $\tilde{v}_I(t)$ can be obtained from the original signal v(t) by multiplying v(t) by a reference sinusoidal signal and then low-passing filtering to remove the high-frequency component. This method is known as the signaldemodulation method, or the phase sensitive detection (PSD) method [15]. Using $2\cos(\omega_0 t)$ and $-2\sin(\omega_0 t)$ as reference signals, signal-demodulation yields $\tilde{v}_R(t)$ and $\tilde{v}_I(t)$, respectively. This detection scheme is known as quadrature detection. Quadrature detection thus produces two data streams with a 90 degrees phase difference. When put in complex form, with $\tilde{v}_R(t)$ being treated as the real part and $\tilde{v}_I(t)$ as the imaginary part, these data streams together constitute the complex envelope $\tilde{v}(t)$ of v(t). Note that, given ω_0 , all information content of v(t) is preserved in the complex envelope $\tilde{v}(t)$.

To get more insight in the relation between v(t) and its quadrature and in-phase components, consider the signal

$$v_{+}(t) = \tilde{v}(t) \exp(\imath \omega_0 t), \tag{16}$$

which is known as the *analytic signal* (or pre-envelope) of v(t) and can be written as

$$v_{+}(t) = v(t) + i\check{v}(t),$$
 (17)

with $\check{v}(t)$ the Hilbert transform of v(t) [16], which will be denoted as $\check{v}(t) = \mathcal{H}[v(t)]$. In other words, $\check{v}(t)$ is the response of a dynamic system with impulse response function

$$h(t) = \frac{1}{\pi t} \tag{18}$$

and corresponding frequency response function

$$H(\iota\omega) = -\iota \operatorname{sgn}(\omega),\tag{19}$$

with $\operatorname{sgn}(\omega)$ the sign of ω . The filter (19), which is known as a quadratic filter [16], has a constant amplitude $|H(\iota\omega)| = 1$ (all pass), and its phase equals $-\pi/2$ for $\omega > 0$ and $\pi/2$ for $\omega < 0$. The effect of forming the complex signal $v_+(t)$ is to remove the redundant negative frequency components of the Fourier transform. Indeed, it follows from above that the Fourier transform $\check{V}(\omega)$ of $\check{v}(t)$ is given by

$$\check{V}(\omega) = -\imath \operatorname{sgn}(\omega) V(\omega), \qquad (20)$$

with $V(\omega)$ the Fourier transform of v(t) and, as follows from (17) and (20),

$$V_{+}(\omega) = V(\omega) + \operatorname{sgn}(\omega)V(\omega).$$
(21)

Furthermore, the Fourier transform of the complex envelope $\tilde{v}(t)$ is given by

$$\tilde{V}(\omega) = V_{+}(\omega + \omega_{0}).$$
(22)

Using the Hilbert transform pairs $\mathcal{H}[\cos(\omega t)] = \sin(\omega t)$ and $\mathcal{H}[\sin(\omega t)] = -\cos(\omega t)$, and Bedrossian's theorem [20] it can be shown that:

$$\tilde{v}_R(t) = v(t)\cos(\omega_0 t) + \check{v}(t)\sin(\omega_0 t), \qquad (23)$$

$$\tilde{v}_I(t) = \check{v}(t)\cos(\omega_0 t) - v(t)\sin(\omega_0 t).$$
(24)

Finally, if we assume that the receiver coil has a homogeneous reception field, which may, for example, be a valid assumption in a single coil based on a birdcage volume resonator [15], the signal expression (15) can be further simplified to

$$\tilde{v}(t) = \int_{object} \underline{M}_{xy}(\underline{r}, t_{acq}) e^{-\imath(\Delta\omega(\underline{r})t + \phi_{pe}(\underline{r}))} d\underline{r}, \qquad (25)$$

where the complex notation

$$\underline{M}_{xy}(\underline{r}, t_{acq}) = |\underline{M}_{xy}(\underline{r}, t_{acq})| e^{i\phi_e(\underline{r})}, \qquad (26)$$

has been used and a scaling constant $= \omega_0 e^{i\pi/2}$ has been omitted [15]. Substituting (3) and (5) in (25) then yields

$$\tilde{v}(\underline{k}) = \int_{object} \underline{M}_{xy}(\underline{r}, t_{acq}) e^{-i2\pi \underline{k} \cdot \underline{r}} d\underline{r}, \qquad (27)$$

where the mapping relation between (t, G_y) and $\underline{k} = (k_x, k_y)^T$ is given by

$$k_x = \frac{1}{2\pi} \gamma G_x t, \tag{28}$$

$$k_y = \frac{1}{2\pi} \gamma G_y T_{pe}. \tag{29}$$

Hence, the signal $\tilde{v}(\underline{k})$ in the so-called k-space is the 2D spatial Fourier transform of $\underline{M}_{xy}(\underline{r}, t_{acq})$, which is the quantity of interest to be reconstructed (i.e., the *image*). Note that if $\phi_e(\underline{r})$ is small or (ideally) zero, the imaginary component of $\underline{M}_{xy}(\underline{r}, t_{acq})$ can be neglected making the image to be reconstructed real valued. In practice, however, the realness constraint is often violated because object motion and magnetic field inhomogeneities introduce a nonzero phase to the image function [15]. Obviously, a straightforward reconstruction of the image is obtained by inverse Fourier transforming the data, but before we come to that, we will first consider the effect of noise. It should be noted that the assumption of a homogeneous reception field is generally invalid for single coil acquisitions that use so-called surface coils [21]. In that case, the detection sensitivity has to be taken into account and (15) can no longer be simplified to (25).

2.2. Modeling the noise

So far, we have considered noiseless signals. In practice, however, the signal v(t) will be disturbed by noise, which is mainly caused by thermal motion (Brownian motion) of electrons within the body's conducting tissue and the receiving coil(s) [22]. This thermal noise, which was investigated experimentally by Johnson [23] and theoretically by Nyquist [24], is often referred to as Johnson noise. It can be modeled as additive zero mean white Gaussian noise with variance (or, power) [25][26]

$$\sigma_w^2 = 2k_b T (R_{coil} + R_{body}), \tag{30}$$

where k_b denotes Boltzmann's constant, T the absolute temperature and R_{coil} and R_{body} the effective resistance of the coil and the body, respectively. Hence, the raw MR signal can be modeled as:

$$\boldsymbol{v}(t) = \boldsymbol{v}(t) + \boldsymbol{n}_w(t), \tag{31}$$

with $\mathbf{n}_w(t)$ a stationary zero mean Gaussian white noise process with variance σ_w^2 . Note that the signal v(t) is bandlimited to frequencies ω such that $||\omega| - \omega_0| \leq \max_{\underline{r}} \Delta \omega(\underline{r})$. The noise, however, has a spectrum that exists over the entire frequency range and can be separated into two components: (i) the out-of-band noise component $\mathbf{n}_o(t)$ and (ii) the in-band noise component $\mathbf{n}(t)$ [17]:

$$\mathbf{n}_w(t) = \mathbf{n}_o(t) + \mathbf{n}(t). \tag{32}$$

The in-band noise component $\mathbf{n}(t)$ can be obtained by filtering the raw data by an (ideal) bandpass filter with a passband that corresponds with the bandpass signal v(t). Note that this filter leaves the bandpass signal v(t)unaffected. Furthermore, note that $\mathbf{n}(t)$ is obtained from a Gaussian process through a linear operation. Hence, the process $\mathbf{n}(t)$ is also a Gaussian process. It is commonly known as a bandpass "white" Gaussian noise process, having a power spectral density function that is symmetrical about ω_0 . It can be shown that $\mathbf{n}_0(t)$ is independent of both v(t) and $\mathbf{n}(t)$ and can be discarded without any loss of information [17]. In what follows, we will assume that bandpass filtering has been applied to eliminate the out of band noise component.

Now, it can be shown that the band pass white Gaussian noise process $\mathbf{n}(t)$ can be written in the form [16]

$$\mathbf{n}(t) = \tilde{\boldsymbol{n}}_R(t)\cos(\omega_0 t) - \tilde{\boldsymbol{n}}_I(t)\sin(\omega_0 t), \tag{33}$$

with $\tilde{\boldsymbol{n}}_R(t)$ and $\tilde{\boldsymbol{n}}_I(t)$ zero mean lowpass stationary Gaussian processes described by

$$\tilde{\boldsymbol{n}}_R(t) = \mathbf{n}(t)\cos(\omega_0 t) + \check{\boldsymbol{n}}(t)\sin(\omega_0 t), \qquad (34)$$

$$\tilde{\boldsymbol{n}}_{I}(t) = \check{\mathbf{n}}(t)\cos(\omega_{0}t) - \boldsymbol{n}(t)\sin(\omega_{0}t)$$
(35)

with $\check{\boldsymbol{n}}(t)$ the Hilbert transform of $\boldsymbol{n}(t)$, where the Hilbert transform $\check{\mathbf{x}}(t)$ of a stochastic process $\mathbf{x}(t)$ is given by the output of the system (19) with input $\mathbf{x}(t)$, that is [16]

$$\check{\mathbf{x}}(t) = \frac{1}{\pi t} * \mathbf{x}(t) = \frac{1}{\pi} \int_{-\infty}^{\infty} \frac{\mathbf{x}(\tau)}{t - \tau} d\tau.$$
(36)

with * the convolution operator. Note the analogy of (34) and (35) with (23) and (24). It can easily be shown that since $\mathbf{n}(t)$ is a stationary process, the process $\check{\mathbf{n}}(t)$ is also stationary [16]. Furthermore, it can be been shown that the following relations hold [16]:

$$R_{\mathbf{\check{n}n}}(0) = 0, \tag{37}$$

$$R_{\mathbf{n}}(\tau) = R_{\tilde{\boldsymbol{n}}_R}(\tau) \cos(\omega_0 \tau), \qquad (38)$$

$$R_{\tilde{\boldsymbol{n}}_R}(\tau) = R_{\tilde{\boldsymbol{n}}_I}(\tau) = R_{\mathbf{n}}(\tau)\cos(\omega_0\tau) + R_{\tilde{\mathbf{n}}\mathbf{n}}(\tau)\sin(\omega_0\tau), \qquad (39)$$

$$R_{\tilde{\boldsymbol{n}}_R\tilde{\boldsymbol{n}}_I}(\tau) = -R_{\tilde{\boldsymbol{n}}_I\tilde{\boldsymbol{n}}_R}(\tau) = 0, \quad \forall \tau$$

$$\tag{40}$$

with $R_{\mathbf{\tilde{n}n}}(\tau)$ the cross-correlation function between the processes $\mathbf{\check{n}}(t)$ and $\mathbf{n}(t)$, $R_{\mathbf{n}}(\tau)$ the autocorrelation function of the process $\mathbf{n}(t)$, $R_{\mathbf{\tilde{n}}_R}(\tau)$ the autocorrelation function of the process $\mathbf{\tilde{n}}_R(t)$, $R_{\mathbf{\tilde{n}}_I}(\tau)$ the autocorrelation function of the process $\mathbf{\tilde{n}}_I(t)$, and $R_{\mathbf{\tilde{n}}_R\mathbf{\tilde{n}}_I}(\tau)$ the cross-correlation function of the processes $\mathbf{\tilde{n}}_R(t)$ and $\mathbf{\tilde{n}}_I(t)$. It follows from (37) that, for a given t, the zero mean Gaussian random variables $\mathbf{n}(t)$ and $\mathbf{\check{n}}(t)$ are orthogonal (and thus

uncorrelated) and it follows from (39) and (40) that the zero mean processes $\tilde{\boldsymbol{n}}_R(t)$ and $\tilde{\boldsymbol{n}}_I(t)$ have equal autocorrelation functions and are orthogonal.

In analogy with (9) and (17), we can define the complex processes

$$\tilde{\mathbf{n}}(t) = \tilde{\boldsymbol{n}}_R(t) + \imath \tilde{\boldsymbol{n}}_I(t), \qquad (41)$$

and

$$\mathbf{n}_{+}(t) = \mathbf{n}(t) + \imath \check{\mathbf{n}}(t) = \tilde{\mathbf{n}}(t) \exp(\imath \omega_{0} t), \qquad (42)$$

representing the complex envelope and the analytic signal associated with n(t). Note that the last equality in (42) follows from Eqs. (34-35).

2.3. Modeling the noise disturbed MR signal

Let's now combine the results obtained in the subsections 2.1 and 2.2 and define

$$\boldsymbol{v}_{+}(t) = \tilde{\boldsymbol{v}}(t) \exp(\imath \omega_0 t), \tag{43}$$

with

$$\tilde{\boldsymbol{v}}(t) = \tilde{\boldsymbol{v}}(t) + \tilde{\mathbf{n}}(t), \tag{44}$$

where $\tilde{v}(t)$ and $\tilde{\mathbf{n}}(t)$ are the previously defined *complex envelopes* of the signal v(t) and the noise process $\mathbf{n}(t)$, respectively. The signal $\tilde{v}(t)$ thus represents the complex envelope of the noise disturbed signal $v(t) + \mathbf{n}(t)$. It follows from above that $\tilde{v}(t)$ can be described as

$$\tilde{\boldsymbol{v}}(t) = \tilde{\boldsymbol{v}}_R(t) + \imath \tilde{\boldsymbol{v}}_I(t), \tag{45}$$

with

$$\tilde{\boldsymbol{v}}_R(t) = \tilde{v}_R(t) + \tilde{\boldsymbol{n}}_R(t), \qquad (46)$$

the noise disturbed in-phase component and

$$\tilde{\boldsymbol{v}}_I(t) = \tilde{\boldsymbol{v}}_I(t) + \tilde{\boldsymbol{n}}_I(t), \qquad (47)$$

the noise disturbed quadrature component. It can be shown that the signals $\tilde{\boldsymbol{v}}_R(t)$ and $\tilde{\boldsymbol{v}}_I(t)$ can be obtained by applying the signal demodulation method described in subsection 2.1, the noise contributions to the in-phase and quadrature detection channel being equal to $\tilde{\boldsymbol{n}}_R(t)$ and $\tilde{\boldsymbol{n}}_I(t)$, respectively.

In summary, the noise disturbed MR data obtained by quadrature detection using a single receiver coil can be described in complex form as:

$$\tilde{\boldsymbol{v}}(t) = \tilde{v}_R(t) + \tilde{\boldsymbol{n}}_R(t) + \imath [\tilde{v}_I(t) + \tilde{\boldsymbol{n}}_I(t)].$$
(48)

where $\tilde{v}_R(t)$ and $\tilde{v}_I(t)$ are the in-phase (or, real) and quadrature (or, imaginary) components of the noiseless signal and $\tilde{n}_R(t)$ and $\tilde{n}_I(t)$ are two zero mean Gaussian, orthogonal processes that describe the in-phase and quadrature component of the noise, respectively.

2.4. Sampling

In MRI practice, the signal (48) will be sampled, and sampling may affect the correlation properties of the noise. Recall that the noise process $\mathbf{n}(t)$ is assumed to be the result of band-pass filtering a continuous time white Gaussian noise process $\mathbf{n}_w(t)$ over a band centered around ω_0 , where the width W of the band corresponds with the passband of the bandpass signal v(t). Furthermore, recall that the power spectral density function (and thus variance) of $\mathbf{n}_w(t)$ was equal to σ_w^2 . Then, the autocorrelation function of the bandpass "white" noise process $\mathbf{n}(t)$ is given by [19]

$$R_{\boldsymbol{n}}(\tau) = 2\sigma_w^2 \frac{\sin(\frac{W}{2}\tau)}{\pi\tau} \cos(\omega_0 \tau).$$
(49)

By comparison with (38) we have

$$R_{\tilde{\boldsymbol{n}}_R}(\tau) = R_{\tilde{\boldsymbol{n}}_I}(\tau) = 2\sigma_w^2 \frac{\sin(\frac{W}{2}\tau)}{\pi\tau}.$$
(50)

If we sample the complex envelope $\tilde{\mathbf{n}}(t) = \tilde{\boldsymbol{n}}_R(t) + i\tilde{\boldsymbol{n}}_I(t)$ at the Nyquist rate of $\omega_s = \frac{2\pi}{\Delta t} = W$, we have:

$$R_{\tilde{\boldsymbol{n}}_R}(l\Delta t) = R_{\tilde{\boldsymbol{n}}_I}(l\Delta t) = \frac{\sigma_w^2 W}{\pi} \frac{\sin(\pi l)}{\pi l} = \frac{\sigma_w^2 W}{\pi} \delta[l], \qquad (51)$$

with $l \in \mathbb{Z}$ and $\delta[\cdot]$ the Kronecker delta function. Furthermore, as derived earlier,

$$R_{\tilde{\boldsymbol{n}}_R\tilde{\boldsymbol{n}}_I}(\tau) = -R_{\tilde{\boldsymbol{n}}_I\tilde{\boldsymbol{n}}_R}(\tau) = 0.$$
(52)

Hence,

$$R_{\tilde{\boldsymbol{n}}_R}[l] = R_{\tilde{\boldsymbol{n}}_I}[l] = \frac{\sigma_w^2 W}{\pi} \delta[l], \qquad (53)$$

and

$$R_{\tilde{\boldsymbol{n}}_R\tilde{\boldsymbol{n}}_I}[l] = -R_{\tilde{\boldsymbol{n}}_I\tilde{\boldsymbol{n}}_R}[l] = 0,$$
(54)

where $R_{\tilde{\boldsymbol{n}}_R}[l] = R_{\tilde{\boldsymbol{n}}_R}(l\Delta t), R_{\tilde{\boldsymbol{n}}_I}[l] = R_{\tilde{\boldsymbol{n}}_I}(l\Delta t), R_{\tilde{\boldsymbol{n}}_R\tilde{\boldsymbol{n}}_I}[l] = R_{\tilde{\boldsymbol{n}}_R\tilde{\boldsymbol{n}}_I}(l\Delta t)$ and $R_{\tilde{\boldsymbol{n}}_I\tilde{\boldsymbol{n}}_R}[l] = R_{\tilde{\boldsymbol{n}}_I\tilde{\boldsymbol{n}}_R}(l\Delta t)$. Hence, the discrete random process obtained by sampling $\tilde{\boldsymbol{n}}(t)$ at the Nyquist rate is complex white Gaussian noise [19].

It's worthwhile mentioning that, assuming frequency encoding along the x-direction, it follows from (3) that the bandwidth W is directly related to the field of view in the x-direction:

$$W = 2 \max_{\underline{r}} |\Delta\omega(\underline{r})| = 2\gamma \max_{x} |G_x x| = \gamma |G_x| FOV_x,$$
(55)

with FOV_x the field of view in the x-direction.

2.5. Modeling k-space data

It follows from (25) and (28-29) that, for a given value of G_y , sampling the signal $\tilde{\boldsymbol{v}}(t)$ at sample intervals Δt corresponds with sampling the k-space along a line of constant k_y at sample intervals $\Delta k_x = \frac{1}{2\pi}\gamma G_x \Delta t$. Assuming a rectilinear sampling scheme, $\tilde{\boldsymbol{v}}(t, G_y)$ is sampled line by line, and the two-dimensional sampling problem can be treated along each dimension separately. Using the mapping relations (28) and (29), it can be shown that the largest sampling intervals permissible by the Nyquist criterion are

$$\Delta t = \frac{2\pi}{\gamma |G_x| FOV_x} \tag{56}$$

and

$$\Delta G_y = \frac{2\pi}{\gamma T_{pe} FOV_y},\tag{57}$$

with FOV_y the field of view in the y-direction [15].

Finally, let $\boldsymbol{z}(\underline{k})$ denote the signal $\tilde{\boldsymbol{v}}(t, G_y)$ that has been mapped to the *k*-space (i.e., spatial frequency space), using the mapping relations (28) and (29):

$$\boldsymbol{z}(\underline{k}) \equiv \tilde{\boldsymbol{v}}(\underline{k}) = \tilde{\boldsymbol{v}}_R(\underline{k}) + \tilde{\boldsymbol{n}}_R(\underline{k}) + \imath [\tilde{\boldsymbol{v}}_I(\underline{k}) + \tilde{\boldsymbol{n}}_I(\underline{k})].$$
(58)

Furthermore, assume that the k-space has been Nyquist sampled in the sample points $\underline{k}_1, \ldots, \underline{k}_N$ and define the complex random vector

$$\underline{\mathbf{z}} = \begin{pmatrix} \mathbf{z}(\underline{k}_1) \\ \vdots \\ \mathbf{z}(\underline{k}_N) \end{pmatrix}$$
(59)

and the complex deterministic vector

$$\underline{s} = \begin{pmatrix} s(\underline{k}_1) \\ \vdots \\ s(\underline{k}_N) \end{pmatrix}, \tag{60}$$

with $s(\underline{k}_i) = s_R(\underline{k}_i) + is_I(\underline{k}_i)$, for i = 1, ..., N. It follows from the analysis described in subsection 2.4 that the real and imaginary parts $\mathbf{z}_R(\underline{k}_i)$ and $\mathbf{z}_I(\underline{k}_i)$ of the complex random variables $\mathbf{z}(\underline{k}_i)$ are independent, Gaussian distributed with equal variance $\sigma_K^2 = \frac{\sigma_w^2 W}{\pi}$, which implies that

$$cov(\mathbf{z}_R(\underline{k}_i), \mathbf{z}_I(\underline{k}_j)) = 0, \forall i, j$$
(61)

and

$$cov(\mathbf{z}_R(\underline{k}_i), \mathbf{z}_R(\underline{k}_j)) = cov(\mathbf{z}_I(\underline{k}_i), \mathbf{z}_I(\underline{k}_j)) = \sigma_K^2 \delta[i-j].$$
(62)

Using basic theory on complex Gaussian distributions (see Appendix A), it then follows that the joint probability density function of the complex random variable \underline{z} is given by:

$$f_{\underline{z}}(\underline{z}) = \frac{1}{\pi^N \det(\Sigma_z)} \exp\{-(\underline{z} - \underline{s})^H \Sigma_z^{-1}(\underline{z} - \underline{s})\}, \qquad (63)$$

with $\Sigma_z = 2\sigma_K^2 I_N$, where I_N is the identity matrix of order N. This is usually denoted by

$$\underline{\boldsymbol{z}} \sim \mathcal{CN}(\underline{s}, 2\sigma_K^2 I_N). \tag{64}$$

This PDF is called the joint circularly complex normal distribution, also known as the complex multivariate normal (or Gaussian) PDF [19, 27].

Expression (64) is the main result of this section and forms the starting point of section 3, in which we will analyze how the data distribution changes when \underline{z} is further processed for the purpose of image reconstruction.

3. MRI data distributions of single coil images

From the complex data in the k-space (i.e., the spatial frequency domain) a so-called reconstructed MR image in the spatial domain can be obtained by taking the inverse two-dimensional (2D) Discrete Fourier Transform (DFT). From the complex valued reconstructed image thus obtained, magnitude and phase images can be created straightforwardly. As illustrated in section 2, the pixel values of the magnitude image are directly related to the strength of the transverse component of the net transverse magnetization in the volume elements, voxels, in the selected tissue slice. The phase image is often discarded since it may exhibit incidental phase variations due to RF angle inhomogeneity, filter responses, system delay, noncentred sampling windows, a time-varying behavior due to radio-frequency angle inhomogeneity, system delay, field inhomogeneities, chemical shift, etc. [28–30]. Magnitude images are immune to these effects. Nevertheless, phase images may contain valuable information. For example, phase images are used to measure flow [31–35] or susceptibility [36–40].

In this section, we will describe the distribution of reconstructed complex, magnitude and phase images acquired with a single-coil acquisition system. The distribution of images acquired with multiple coils systems will be described in section 4.

3.1. Statistical distribution of single-coil complex images

As was derived in section 2, the complex data acquired in k-space, often referred to as the *raw data*, can (pixel-wise) be described as

$$\boldsymbol{z}(\underline{k}) = \boldsymbol{s}(\underline{k}) + \boldsymbol{n}(\underline{k}), \tag{65}$$

with $s(\underline{k})$ the complex noise free data and $n(\underline{k})$ additive noise that can be modeled as a zero mean circular complex Gaussian random variable (see Appendix A), whose PDF is given by

$$f_{\boldsymbol{n}(\underline{k})}(n(\underline{k})) = \frac{1}{2\pi\sigma_K^2} \exp\left(-|n(\underline{k})|^2/(2\sigma_K^2)\right).$$
(66)

This is usually denoted as

$$\boldsymbol{n}(\underline{k}) \sim \mathcal{CN}(0, 2\sigma_K^2),$$
 (67)

which implies that the real and imaginary components $\boldsymbol{n}_R(\underline{k})$ and $\boldsymbol{n}_I(\underline{k})$ of $\mathbf{n}(\underline{k})$ are independent identically distributed (i.i.d.) zero-mean Gaussian random variables with variance σ_K^2 (see Appendix A). That is,

$$\boldsymbol{n}_R(\underline{k}) \sim \mathcal{N}(0, \sigma_K^2),$$
 (68)

$$\boldsymbol{n}_I(\underline{k}) \sim \mathcal{N}(0, \sigma_K^2),$$
 (69)

and $cov(\boldsymbol{n}_{R}(\underline{k}), \boldsymbol{n}_{I}(\underline{k})) = 0$. Moreover, it follows from (66) that the RV $\mathbf{n}(\underline{k})$ has the same distribution as the RV $e^{i\theta}\mathbf{n}(\underline{k}), \forall \theta \in \mathbb{R}$. The RV $\mathbf{n}(\underline{k})$ is therefore called *circularly-symmetric* [41].

Furthermore, as explained in section 2, the noise process $n(\underline{k})$ can be assumed to be stationary so that σ_K^2 does not depend on \underline{k} . Moreover, as was shown in section 2, if we assume that the k-space was sampled at the Nyquist rate, the complex Gaussian distributed sample points in the k-space are uncorrelated (and therefore independent).

Next, a complex image in the spatial domain (or, *image space*) is obtained by taking the inverse DFT of the complex data in the k-space. Due to the linearity and orthogonality of the Fourier transform, the complex data points in the image space are also independent Gaussian distributed, as is illustrated in Appendix B. Hence, the complex image in the spatial domain can (pixelwise) be modeled as

$$\boldsymbol{z}(\underline{r}) = \boldsymbol{s}(\underline{r}) + \boldsymbol{n}(\underline{r}), \tag{70}$$

with

$$s(\underline{r}) = s_R(\underline{r}) + \imath s_I(\underline{r}), \tag{71}$$

the noise free signal and

$$\boldsymbol{n}(\underline{r}) = \boldsymbol{n}_R(\underline{r}) + \imath \boldsymbol{n}_I(\underline{r})$$
(72)

the additive noise contribution, and $\mathbf{n}(\underline{r}) \sim \mathcal{CN}(0, 2\sigma^2)$, with $\sigma^2 = \frac{1}{N}\sigma_K^2$, where N is the number of points used to compute the inverse DFT (see Appendix B). This implies that

$$\boldsymbol{n}_R(\underline{r}) \sim \mathcal{N}(0, \sigma^2),$$
 (73)

$$\boldsymbol{n}_I(\underline{r}) \sim \mathcal{N}(0, \sigma^2),$$
 (74)

and

$$cov(\boldsymbol{n}_R(\underline{r}), \boldsymbol{n}_I(\underline{r})) = 0.$$
 (75)

The probability density function (PDF) of the complex Gaussian RV $\boldsymbol{z}(\underline{r})$ is then given by

$$f_{\boldsymbol{z}(\underline{r})}(\boldsymbol{z}(\underline{r})) = \frac{1}{2\pi\sigma^2} \exp\left(-\frac{|\boldsymbol{z}(\underline{r}) - \boldsymbol{s}(\underline{r})|^2}{2\sigma^2}\right),\tag{76}$$

which is usually denoted as $\boldsymbol{z}(\underline{r}) \sim \mathcal{CN}(s(\underline{r}), 2\sigma^2)$.

3.2. Statistical distribution of single-coil magnitude images

A magnitude image is obtained by taking (pixel by pixel) the root sum of squares (SoS) of the real and imaginary part of the complex image $\boldsymbol{z}(\underline{r})$:

$$\mathbf{m}(\underline{r}) = \sqrt{\boldsymbol{z}_R^2(\underline{r}) + \boldsymbol{z}_I^2(\underline{r})} = \sqrt{|\boldsymbol{z}(\underline{r})|^2}.$$
(77)

For notational convenience, we will suppose that all the equations are pixelwise and write \boldsymbol{z} and \boldsymbol{m} instead of $\boldsymbol{z}(\underline{r})$ and $\boldsymbol{m}(\underline{r})$. It can be shown that the random variable \boldsymbol{m} is Rician distributed. Its PDF $f_{\mathbf{m}}(m)$ is given by [42]

$$f_{\mathbf{m}}(m) = \frac{m}{\sigma^2} e^{-\frac{a^2 + m^2}{2\sigma^2}} I_0\left(\frac{ma}{\sigma^2}\right) \epsilon(m), \tag{78}$$

where $I_0(\cdot)$ is the 0th order modified Bessel function of the first kind and $a^2 = s_R^2 + s_I^2$, with s_R and s_I the real and imaginary part of $s = \mathbb{E}[\mathbf{z}]$. The unit step Heaviside function $\epsilon(.)$ is used to indicate that the expression for the PDF of **m** is valid for non-negative values of *m* only.

The shape of the PDF (78) depends on the Signal to Noise Ratio (SNR), which we will define as a/σ . In the special case a = 0 (no signal, SNR=0), the Rician PDF turns into a Rayleigh PDF given by [43, 44]

$$f_{\mathbf{m}}(m) = \frac{m}{\sigma^2} e^{-\frac{m^2}{2\sigma^2}} \epsilon(m).$$
(79)

For increasing values of the SNR, that is, for SNR $\rightarrow \infty$, the asymptotic expansion of $I_0(x)$ when x is large is [45]

$$I_0(x) \sim \frac{e^x}{\sqrt{2\pi x}} \left[1 + \frac{1}{8x} + \frac{1.9}{2!.(8x)^2} + \frac{1.9.25}{2!.(8x)^3} + \cdots \right]$$
(80)

Then, for sufficiently large x, $I_0(x) \approx \frac{e^x}{\sqrt{2\pi x}}$ and the Rician distribution (78) can be approximated as follows:

$$f_{\mathbf{m}}(m) = \sqrt{\frac{m}{2\pi\sigma^2 a}} \exp\left(-\frac{(m-a)^2}{2\sigma^2}\right) \quad , \tag{81}$$

or even further by a Gaussian distribution with corresponding PDF

$$f_{\mathbf{m}}(m) = \frac{1}{\sigma\sqrt{2\pi}} \exp\left(-\frac{(m-a)^2}{2\sigma^2}\right).$$
(82)

The moments (or raw moments) of the Rician distribution can be expressed analytically as [46]

$$\mathbb{E}[\mathbf{m}^r] = \left(2\sigma^2\right)^{r/2} \Gamma\left(1 + \frac{r}{2}\right) {}_1F_1\left[-\frac{r}{2}; 1; -\frac{a^2}{2\sigma^2}\right],\tag{83}$$

where $\Gamma(\cdot)$ represents the Gamma function and ${}_{1}F_{1}(\cdot; \cdot; \cdot)$ denotes the confluent hypergeometric function of the first kind. The first four moments of the Rice PDF are given by [47]

$$\mathbb{E}\left[\mathbf{m}\right] = \sigma \sqrt{\frac{\pi}{2}} e^{-\frac{a^2}{4\sigma^2}} \left[\left(1 + \frac{a^2}{2\sigma^2}\right) I_0\left(\frac{a^2}{4\sigma^2}\right) + \frac{a^2}{2\sigma^2} I_1\left(\frac{a^2}{4\sigma^2}\right) \right], \quad (84)$$
$$\mathbb{E}\left[\mathbf{m}^2\right] = a^2 + 2\sigma^2, \quad (85)$$

$$\mathbb{E}\left[\mathbf{m}^{3}\right] = \sigma^{3}\sqrt{\frac{\pi}{2}}e^{-\frac{a^{2}}{4\sigma^{2}}}\left[\left(3+3\frac{a^{2}}{\sigma^{2}}+\frac{a^{4}}{2\sigma^{4}}\right)I_{0}\left(\frac{a^{2}}{4\sigma^{2}}\right)\right.$$
$$\left.+\left(2\frac{a^{2}}{\sigma^{2}}+\frac{a^{4}}{2\sigma^{4}}\right)I_{1}\left(\frac{a^{2}}{4\sigma^{2}}\right)\right],\tag{86}$$
$$\mathbb{E}\left[\mathbf{m}^{4}\right] = \frac{4+9}{\sigma^{2}}e^{-\frac{2}{4\sigma^{2}}}+9,4$$

$$\mathbb{E}\left[\mathbf{m}^{4}\right] = a^{4} + 8\sigma^{2}a^{2} + 8\sigma^{4}, \tag{87}$$

with $I_1(\cdot)$ denoting the 1st order modified Bessel function of the first kind. Note that the even moments are simple polynomials. The expressions for the odd moments are more complex and have been derived using the fact that the confluent hypergeometric function can be expressed in terms of modified Bessel functions [45]. The variance of the Rician distributed RV **m** is given by

$$Var(\mathbf{m}) = \mathbb{E}[\mathbf{m}^{2}] - \mathbb{E}[\mathbf{m}]^{2}$$

$$= a^{2} + 2\sigma^{2}$$

$$-\frac{\pi\sigma^{2}}{2}e^{-\frac{a^{2}}{2\sigma^{2}}} \left[\left(1 + \frac{a^{2}}{2\sigma^{2}}\right) I_{0}\left(\frac{a^{2}}{4\sigma^{2}}\right) + \frac{a^{2}}{2\sigma^{2}}I_{1}\left(\frac{a^{2}}{4\sigma^{2}}\right) \right]^{2}.$$
(88)

Since the Rayleigh PDF is a special case of the Rice PDF (with a = 0), expressions for its moments can be directly derived from the expressions above, yielding [48]

$$\mathbb{E}\left[\mathbf{m}^{r}\right] = (2\sigma^{2})^{r/2}\Gamma\left(1+\frac{r}{2}\right),\tag{89}$$

or explicitly for the first four moments:

$$\mathbb{E}[\mathbf{m}] = \sqrt{\frac{\pi}{2}}\sigma, \tag{90}$$

$$\mathbb{E}[\mathbf{m}^2] = 2\sigma^2, \tag{91}$$

$$\mathbb{E}[\mathbf{m}^3] = 3\sqrt{\frac{\pi}{2}}\sigma^3, \tag{92}$$

$$\mathbb{E}[\mathbf{m}^4] = 8\sigma^4, \tag{93}$$

and the variance (88) simplifies to

$$Var(\mathbf{m}) = \sigma^2 \left(2 - \frac{\pi}{2}\right). \tag{94}$$

At high SNR, on the other hand, the Rician PDF tends to a Gaussian PDF and (88) will reduce to the simple expression

$$Var(\mathbf{m}) = \sigma^2. \tag{95}$$

3.3. Statistical distribution of single-coil phase images

Recall from subsection 3.1 that the PDF of the complex Gaussian RV zwith mean s and variance $2\sigma^2$ is given by

$$f_{\boldsymbol{z}}(z) = \frac{1}{2\pi\sigma^2} \exp\left(-\frac{|z-s|^2}{2\sigma^2}\right),\tag{96}$$

which is usually denoted as $\boldsymbol{z} \sim \mathcal{CN}(s, 2\sigma^2)$. If we write \boldsymbol{z} in polar coordinates, we get $\boldsymbol{z} = \boldsymbol{m}e^{i\boldsymbol{\theta}}$, where the real valued RVs \boldsymbol{m} and $\boldsymbol{\theta}$ denote the magnitude and phase of \boldsymbol{z} , respectively. Similarly, the complex mean s can also be written in polar coordinates: $s = ae^{i\phi}$. The joint PDF of \boldsymbol{m} and $\boldsymbol{\theta}$ is obtained by rewriting (96) as

$$f_{\mathbf{m},\theta}(m,\theta) = \frac{1}{2\pi\sigma^2} \exp\left(-\frac{|me^{i\theta} - ae^{i\phi}|^2}{2\sigma^2}\right).$$
(97)

As discussed above, the magnitude **m** is Rician distributed. Its PDF is given by (78). The conditional distribution of the phase $\boldsymbol{\theta}$, given the magnitude, follows a so-called Tikhonov distribution [49] [50]:

$$f_{\theta|m}(\theta|m) = \frac{\exp\left[\lambda\cos(\theta - \phi)\right]}{2\pi I_0(\lambda)},\tag{98}$$

with $\lambda = \frac{ma}{\sigma^2}$. It is obtained by dividing (97) by (78). The marginal PDF of the phase $\boldsymbol{\theta}$ is obtained by integrating (97) over *m* yielding [51–53]

$$f_{\theta}(\theta) = \frac{1}{2\pi} \exp\left[-\frac{1}{2} \left(\frac{a}{\sigma}\right)^2\right] \times \left[1 + \kappa \sqrt{\pi} \exp(\kappa^2)(1 + erf(\kappa))\right], \quad (99)$$

with $erf(\cdot)$ the error function

$$erf(x) = \frac{2}{\sqrt{\pi}} \int_0^x e^{-t^2} dt,$$
 (100)

and

$$\kappa = \frac{1}{\sqrt{2}} \frac{a}{\sigma} \cos(\theta - \phi). \tag{101}$$

Note that, unlike \mathbf{z}_R and \mathbf{z}_I , the RVs \mathbf{m} and $\boldsymbol{\theta}$ are generally not independent. However, when a = 0, (98) and (99) reduce to a uniform PDF $f_{\boldsymbol{\theta}|\boldsymbol{m}}(\boldsymbol{\theta}|\boldsymbol{m}) = f_{\boldsymbol{\theta}}(\boldsymbol{\theta}) = \frac{1}{2\pi}$ and \mathbf{m} and $\boldsymbol{\theta}$ are independent. For high SNR on the other hand, (99) tends to the Gaussian PDF [53]

$$f_{\theta}(\theta) = \frac{1}{\sqrt{2\pi}} \frac{a}{\sigma} \exp\left[-\frac{a^2(\theta-\phi)^2}{2\sigma^2}\right].$$
 (102)

3.4. Simplifications at high SNR

Note that in polar coordinates, the complex images can (pixel-wise) be described as

$$\mathbf{z}(\underline{r}) = \mathbf{m}(\underline{r})e^{\imath\boldsymbol{\theta}(\underline{r})} = a(\underline{r})e^{\imath\phi(\underline{r})} + |\boldsymbol{n}(\underline{r})|e^{\imath\boldsymbol{\psi}(\underline{r})}$$
(103)

with

$$|\boldsymbol{n}(\underline{r})| = \sqrt{\boldsymbol{n}_R^2(\underline{r}) + \boldsymbol{n}_I^2(\underline{r})}, \qquad (104)$$

the Rayleigh distributed magnitude of the noise and

$$\boldsymbol{\psi}(\underline{r}) = \tan^{-1} \left(\frac{\boldsymbol{n}_I(\underline{r})}{\boldsymbol{n}_R(\underline{r})} \right), \qquad (105)$$

the uniformly distributed phase of the noise. The random variables $|\boldsymbol{n}(\underline{r})|$ and $\boldsymbol{\psi}(\underline{r})$ are independent and their statistical properties do not depend on \underline{r} . From now on, the dependence of $\mathbf{z}, \mathbf{m}, a, \phi, \boldsymbol{n}$ and $\boldsymbol{\psi}$ on \underline{r} is assumed but dropped from the notation. Expression (103) can be rewritten as

$$\mathbf{z} = e^{i\phi} \left(a + |\mathbf{n}| e^{i(\boldsymbol{\psi} - \phi)} \right) = e^{i\phi} \left(a + |\mathbf{n}| \cos(\boldsymbol{\psi} - \phi) + i|\mathbf{n}| \sin(\boldsymbol{\psi} - \phi) \right), \quad (106)$$

with $|\mathbf{n}| \cos(\boldsymbol{\psi} - \phi)$ the noise component that is collinear (i.e., in phase) with the signal and $|\mathbf{n}| \sin(\boldsymbol{\psi} - \phi)$ the noise component that is out of phase with the signal. It follows from (106) that

$$\boldsymbol{m} \equiv |\mathbf{z}| = |a + |\boldsymbol{n}|\cos(\boldsymbol{\psi} - \boldsymbol{\phi}) + \imath |\boldsymbol{n}|\sin(\boldsymbol{\psi} - \boldsymbol{\phi})|.$$
(107)

Starting from (103) and assuming that $\boldsymbol{\psi}$ is distributed uniformly, independent of the value of $|\boldsymbol{n}|$, Hayes and Roemer [54] analyzed the variance of \boldsymbol{m} at high SNR by replacing squares and square roots by power series expansions to second order in $\frac{|\boldsymbol{n}|}{a}$, resulting in

$$Var(\mathbf{m}) \simeq \frac{\mathbb{E}[|\boldsymbol{n}|^2]}{2}.$$
 (108)

Hayes and Roemer [54] found that the approximation (108) is equivalent to ignoring the noise that is out of phase with the signal, that is, by ignoring the imaginary part of the last term in (106). Indeed it can be shown straightforwardly that

$$Var\left(a + |\boldsymbol{n}|\cos(\boldsymbol{\psi} - \boldsymbol{\phi})\right) = \frac{\mathbb{E}[|\boldsymbol{n}|^2]}{2}.$$
 (109)

Hence, one may conclude that in the high SNR case (i.e., $SNR \ge 10$), only the noise in-phase with the signal will contribute to the magnitude image, whereas the contribution of the out-of phase noise can be neglected [54, 55].

Assuming $\mathbf{n} \sim \mathcal{CN}(0, 2\sigma^2)$, $|\mathbf{n}|$ is Rayleigh distributed and (108) reduces to

$$Var(\mathbf{m}) \cong \frac{2\sigma^2}{2} = \sigma^2.$$
 (110)

Note that this observation is in agreement with the earlier mentioned result that for high SNR the distribution of **m** tends to $\mathcal{N}(a, \sigma^2)$. Furthermore, remark that if the signal $ae^{i\phi}$ to be reconstructed (i.e., the *image*) is known to be real-valued and positive (i.e., $\phi = 0$), expression (107) reduces to

$$\boldsymbol{m} = |\mathbf{z}| = |a + |\boldsymbol{n}|\cos(\boldsymbol{\psi}) + i|\boldsymbol{n}|\sin(\boldsymbol{\psi})|, \qquad (111)$$

which can be rewritten as

$$\boldsymbol{m} = |\mathbf{z}| = |a + \mathbf{n}_R + \imath \mathbf{n}_I| \tag{112}$$

In this case, the out of phase noise component corresponds with the imaginary part of the noise. It now follows from the analysis above, that for high SNR and $\phi = 0$, one may reasonably assume that one is dealing solely with the real part of the noise as is often practiced [55]. Note that this assumption will no longer be valid at low(er) SNR. Furthermore, the realness assumption is often violated because object motion and magnetic field inhomogeneities introduce a nonzero phase ϕ to the images.

4. Statistical distribution of multiple-coil images

So far, we have assumed that the images are acquired with a single receiver coil. However, image acquisition with multiple coils is becoming more and more common nowadays. Therefore, this section considers the distribution of MR images acquired by multiple-coil systems.

Before we continue, it should be mentioned that parallel MRI (pMRI) methods are outside the scope of this paper. pMRI allows reducing the acquisition time by sub-sampling the k-space, at the expense of aliasing and other artifacts in the image space. As a consequence, SoS can no longer be used as reconstruction method. Reconstruction methods such as sensitivity encoding (SENSE) and GeneRalized Autocalibrated Partially Parallel Acquisition (GRAPPA) have been introduced to suppress or correct these artifacts. For a review of pMRI methods , the reader is referred to [56]. Furthermore, for an analysis of the noise in GRAPPA and SENSE reconstructed images, see [57] and [58], respectively. Generally, the distribution of images acquired by pMRI methods is still a subject of current research.

4.1. Statistical distribution of multiple-coil complex images

When images are acquired with multiple (say L) receiver coils, the kspace is effectively sampled L times, resulting in L sets of complex raw data. Taking the inverse DFT of each of these data sets results in L complex images in the image space. We have seen above, that the pixels of each of these complex images can be modeled as circularly complex Gaussian random variables (i.e., as complex random variables whose real and imaginary parts are independent, Gaussian distributed with equal variance):

$$\boldsymbol{z}_{l}(\underline{r}) \sim \mathcal{CN}(s_{l}(\underline{r}), 2\sigma_{l}^{2}), \quad l = 1, \dots, L$$
 (113)

with $s_l(\underline{r})$ the expected value and $2\sigma_l^2$ the variance of the pixels of the complex image acquired with the l^{th} coil. Note that the variance of the real and imaginary part of $\boldsymbol{z}_l(\underline{r})$ is given by σ_l^2 . Define (for each \underline{r}) the complex random vector

$$\underline{\mathbf{z}}(\underline{r}) = \begin{pmatrix} \mathbf{z}_1(\underline{r}) \\ \vdots \\ \mathbf{z}_L(\underline{r}) \end{pmatrix}, \qquad (114)$$

and the complex deterministic vector

$$\underline{s}(\underline{r}) = \begin{pmatrix} s_1(\underline{r}) \\ \vdots \\ s_L(\underline{r}) \end{pmatrix}.$$
(115)

In what follows, we will write $\underline{\mathbf{z}}$ and \underline{s} instead of $\underline{\mathbf{z}}(\underline{r})$ and $\underline{s}(\underline{r})$ to simplify the notation. Next, let $\underline{\mathbf{z}}_R$ denote the real part and $\underline{\mathbf{z}}_I$ the imaginary part of $\underline{\mathbf{z}}$ and define

$$\Sigma_R = cov(\underline{\mathbf{z}}_R, \underline{\mathbf{z}}_R), \qquad (116)$$

$$\Sigma_I = cov(\underline{\mathbf{z}}_I, \underline{\mathbf{z}}_I), \qquad (117)$$

$$\Sigma_{IR} = \Sigma_{RI}^T = cov(\underline{\mathbf{z}}_R, \underline{\mathbf{z}}_I), \qquad (118)$$

and

$$\underline{\boldsymbol{w}} = \begin{pmatrix} \underline{\mathbf{z}} \\ \underline{\mathbf{z}}^* \end{pmatrix}. \tag{119}$$

Then it can be shown that the PDF of $\underline{\mathbf{z}}$ is given by (see Appendix A) [59]

$$\frac{1}{\pi^L \sqrt{\det(\Sigma_w)}} \exp\left\{-\frac{1}{2}(\underline{w} - \mathbb{E}[\underline{\mathbf{w}}])^H \Sigma_w^{-1}(\underline{w} - \mathbb{E}[\underline{\mathbf{w}}])\right\}$$
(120)

where the elements of \underline{w} correspond with those of $\underline{\mathbf{w}}$ and $\Sigma_w = cov(\mathbf{w}, \mathbf{w})$.

Next, suppose that we can assume that the correlation structure of the real parts of the noise at the different coils is equal to the correlation structure of the imaginary parts, that is,

$$\Sigma_R = \Sigma_I. \tag{121}$$

Furthermore, let's suppose that we can assume that

$$\Sigma_{IR} = -\Sigma_{IR}^T. \tag{122}$$

If condition (121) and condition (122) are both satisfied, (120) simplifies to a so-called joint circularly complex normal distribution [59], also known as the complex multivariate Gaussian PDF [19] (see Appendix A):

$$f_{\underline{z}}(\underline{z}) = \frac{1}{\pi^L \det(\Sigma_z)} \exp\{-(\underline{z} - \underline{s})^H \Sigma_z^{-1}(\underline{z} - \underline{s})\}, \qquad (123)$$

where the elements of the vector \underline{z} correspond with those of \underline{z} and $\Sigma_z = cov(\underline{z}, \underline{z}) = 2\Sigma_R$. This is usually denoted as $\underline{z} \sim C\mathcal{N}(\underline{s}, 2\Sigma_R)$.

Note that condition (122) implies that Σ_{IR} has a zero main diagonal. This means that the real and imaginary part of each component \mathbf{z}_k of \mathbf{z} are uncorrelated, which is a valid assumption, as was derived in section 2. Furthermore, a sufficient, but not necessary, condition for (122) to be satisfied is

$$\Sigma_{IR} = O, \tag{124}$$

with O the $L \times L$ null matrix. In that case, the real real part of \mathbf{z}_k and the imaginary part of \mathbf{z}_l are uncorrelated not only for k = l, but also for $k \neq l$. This seems to be a reasonable assumption that is often (implicitly) practiced [57, 60].

Moreover, if we not only assume that conditions (121) and (122) are satisfied, but additionally assume that there is no correlation between the coils and that the variance of the noise at each coil is the same, then Σ_R and Σ_I will be diagonal matrices with identical eigenvalues:

$$\Sigma_R = \Sigma_I = \sigma^2 I_L, \tag{125}$$

where I_L is the identity matrix of order L. In this case,

$$\Sigma_z = 2\sigma^2 I_L \tag{126}$$

and (123) further simplifies to

$$f_{\underline{z}}(\underline{z}) = \frac{1}{\pi^L \det(\Sigma_z)} \exp\left\{-\frac{1}{2\sigma^2}(\underline{z}-\underline{s})^H(\underline{z}-\underline{s})\right\} = \frac{1}{(2\pi\sigma^2)^L} e^{-\frac{1}{2\sigma^2}|\underline{z}-\underline{s}|_2^2},$$
(127)

with $|\underline{z} - \underline{s}|_2 = \sqrt{\sum_{l=1}^{L} |z_l - s_l|^2}$ the ℓ^2 -norm of the complex vector $\underline{z} - \underline{s}$. That is, $\underline{z} \sim C\mathcal{N}(\underline{s}, 2\sigma^2 I_L)$. Note that (127) reduces to (76) if L = 1.

4.2. Statistical distribution of multiple-coil magnitude images

When images are acquired with multiple receiver coils and the k-space is fully sampled, a composite magnitude image can be obtained by pixelwise taking the root of the sum of squares (SoS) [61]:

$$\mathbf{m}_{L} = \sqrt{\sum_{l=1}^{L} \left(\mathbf{z}_{R_{l}}^{2} + \mathbf{z}_{I_{l}}^{2} \right)}, \qquad (128)$$

with L the number of coils and \mathbf{z}_{R_l} and \mathbf{z}_{I_l} the real and imaginary component of the complex image obtained from the raw data acquired by the l^{th} coil. Note that we again suppose that all the equations are pixelwise and write \mathbf{m}_L instead of $\mathbf{m}_L(\underline{r})$. If the variance of the noise at each coil is the same and there are no correlations, the PDF of \mathbf{m}_L is given by [62, 63]

$$f_{\mathbf{m}_{L}}(m) = \frac{m^{L}}{\sigma^{2} a^{L-1}} e^{-\frac{m^{2} + a^{2}}{2\sigma^{2}}} I_{L-1}\left(\frac{ma}{\sigma^{2}}\right) \epsilon(m),$$
(129)

where $a^2 = \underline{s}^H \underline{s} = \sum_{l=1}^{L} \left(s_{R_l}^2 + s_{I_l}^2 \right) = \sum_{l=1}^{L} |s_l|^2$, with s_{R_l} and s_{I_l} the means of the real and imaginary components of the complex image pixel values obtained from raw data acquired with the l^{th} coil. The PDF (129) is known as the generalized Rice distribution and is directly related to the so-called non-central chi (nc- χ) distribution. Indeed, it can be shown that the scaled random variable $\mathbf{m}'_L = \frac{\mathbf{m}_L}{\sigma}$, being the root sum of squares of a set of 2Lindependent Gaussian random variables with unit variance, has a nc- χ distribution, with 2L degrees of freedom and non-centrality parameter

$$\lambda = \sqrt{\sum_{l=1}^{L} \left(\left(\frac{s_{R_l}}{\sigma}\right)^2 + \left(\frac{s_{I_l}}{\sigma}\right)^2 \right)} = \frac{a}{\sigma}.$$
 (130)

Its PDF is given by

$$f_{\mathbf{m}_{L}'}(m) = \frac{m^{L}e^{-\frac{m^{2}+\frac{a^{2}}{\sigma^{2}}}{2}}}{\left(\frac{a}{\sigma}\right)^{L-1}}I_{L-1}\left(\frac{ma}{\sigma}\right)\epsilon(m).$$
(131)

It follows from (129) and (131), that $f_{\mathbf{m}_L}(m) = \frac{1}{\sigma} f_{\mathbf{m}'_L}(\frac{m}{\sigma})$. This relation can also be derived directly using basic theory on random variable transformation [64]. In fact, both (129) and (131) are often referred to as nc- χ distributions. In the remainder of this paper, we will follow this convention and refer to (129) as a nc- χ distribution as well.

When $a \rightarrow 0$, the PDF of \mathbf{m}_L turns into a generalized Rayleigh PDF [31, 62]:

$$f_{\mathbf{m}_L}(m) = \frac{2m^{2L-1}}{(\sigma\sqrt{2})^{2L}\Gamma(L)} \exp\left(-\frac{m^2}{2\sigma^2}\right)\epsilon(m), \qquad (132)$$

which can be rewritten as [65]

$$f_{\mathbf{m}_L}(m) = \frac{1}{\Gamma(L)\sigma^2} \left(\frac{m}{2\sigma^2}\right)^{L-1} m^L \exp\left(-\frac{m^2}{2\sigma^2}\right) \epsilon(m).$$
(133)

The moments of the generalized Rice PDF can be expressed analytically as [66]:

$$\mathbb{E}\left[\mathbf{m}_{L}^{r}\right] = (2\sigma^{2})^{r/2} \frac{\Gamma\left[(2L+r)/2\right]}{\Gamma(L)} {}_{1}F_{1}\left(-\frac{r}{2}, L; 3-\frac{a^{2}}{2\sigma^{2}}\right).$$
(134)

or, equivalently [67],

$$\mathbb{E}\left[\mathbf{m}_{L}^{r}\right] = (2\sigma^{2})^{r/2} e^{-\frac{m^{2}+a^{2}}{2\sigma^{2}}} \frac{\Gamma\left[(2L+r)/2\right]}{\Gamma(L)} {}_{1}F_{1}\left(\frac{2L+r}{2}, L; \frac{a^{2}}{2\sigma^{2}}\right), \quad (135)$$

using the transformation $_{1}F_{1}(a, b, z) = e^{z} {}_{1}F_{1}(b - a, b, -z)$ [45]. The mean of \mathbf{m}_{L} is given by [31]

$$\mathbb{E}[\mathbf{m}_L] = \sqrt{2}\sigma \frac{\Gamma(L+\frac{1}{2})}{\Gamma(L)} {}_1F_1\left(-\frac{1}{2}, L; -\frac{a^2}{2\sigma^2}\right).$$
(136)

Again, the even moments turn out to be simple polynomials:

$$\mathbb{E}\left[\mathbf{m}_{L}^{2}\right] = 2L\sigma^{2} + a^{2}, \qquad (137)$$

$$\mathbb{E}\left[\mathbf{m}_{L}^{4}\right] = 4L^{2}\sigma^{4} + 4L\sigma^{4} + 4a^{2}L\sigma^{2} + 4a^{2}\sigma^{2} + a^{4}.$$
 (138)

For a = 0, we obtain the moments of the generalized Rayleigh PDF:

$$\mathbb{E}\left[\mathbf{m}_{L}^{r}\right] = (2\sigma^{2})^{r/2} \frac{\Gamma\left[(2L+r)/2\right]}{\Gamma(L)} \quad , \tag{139}$$

which, using some general properties of the Gamma function $\Gamma(\cdot)$, yields for the mean value and variance [65]

$$\mathbb{E}[\mathbf{m}_{L}] = \frac{1 \cdot 3 \cdot 5 \cdots (2L-1)}{2^{L-1}(L-1)!} \sqrt{\frac{\pi}{2}} \sigma, \qquad (140)$$

and

$$var(\mathbf{m}_{L}) = \mathbb{E}[\mathbf{m}_{L}^{2}] - \mathbb{E}[\mathbf{m}_{L}]^{2} = \left(2L - \left(\frac{1 \cdot 3 \cdot 5 \cdots (2L-1)}{2^{L-1}(L-1)!}\right)^{2}\right) \frac{\pi}{2}\sigma^{2}.$$
(141)

Furthermore, it can be shown that the PDF of the random variable

$$\mathbf{q}_L = \mathbf{m}_L^2 = \sum_{l=1}^L \left(\mathbf{z}_{R_l}^2 + \mathbf{z}_{I_l}^2 \right)$$
(142)

is given by [31]

$$f_{\mathbf{q}_L}(q) = \frac{1}{2\sigma^2} e^{-\frac{a^2+q}{2\sigma^2}} \left(\frac{q}{a^2}\right)^{\frac{L-1}{2}} I_{L-1}\left(\frac{\sqrt{q}a}{\sigma^2}\right) \epsilon(q).$$
(143)

The PDF (143) is directly related to the so-called non-central chi-square PDF. Indeed, it can be shown that the random variable $\mathbf{q}'_L = \frac{\mathbf{q}_L}{\sigma^2}$, being the sum of squares of a set of 2L independent Gaussian random variables with unit variance, has a non-central chi-square (nc- χ^2) distribution, with 2L degrees of freedom and non-centrality parameter $\frac{a^2}{\sigma^2}$. Its PDF is given by [68]

$$f_{\mathbf{q}_{L}'}(q) = \frac{1}{2} e^{-\frac{a^{2}}{\sigma^{2}} + q}} \left(\frac{q\sigma^{2}}{a^{2}}\right)^{\frac{L-1}{2}} I_{L-1}\left(\sqrt{q}\frac{a}{\sigma}\right) \epsilon(q).$$
(144)

It follows from (143) and (144), that $f_{\mathbf{q}_L}(q) = \frac{1}{\sigma^2} f_{\mathbf{q}'_L}(\frac{q}{\sigma^2})$. This relation can also be derived directly using basic theory on random variable transformation [64]. In fact, since (143) is often also referred to as a nc- χ^2 distribution [69][57][60]), we will from now on refer to both (143) and (144) as nc- χ^2 distributions. The variance of \mathbf{m}_L^2 follows directly from Eqs. (137) and (138):

$$var(\mathbf{m}_{L}^{2}) = \mathbb{E}[\mathbf{m}^{4}] - (\mathbb{E}[\mathbf{m}^{2}])^{2} = 4a^{2}\sigma^{2} + 4L\sigma^{4}.$$
 (145)

It is worthwhile mentioning that the generalized Rice distribution also applies to MR data acquired in Phase Contrast Magnetic Resonance (PCMR) imaging, which is a technique that is widely used to detect flow [31, 32, 70].

Recall that to arrive at the generalized Rice distribution (129) and the $nc - \chi^2$ distribution (143), we had to assume that there are no correlations between the coils and the variance of the noise at each coil is the same. In mathematical terms, these assumptions imply that the conditions (124) and (125) should be satisfied¹. However, in phased array (multiple-coil) systems noise correlations may exist [54, 61, 71]. Furthermore, the noise variance may differ from coil to coil. Generally, the noise correlation matrix could be determined experimentally from a reasonably large set of samples reflecting mere noise [60, 72]. Taking noise correlations into account, Aja-Fernandez

¹Strictly speaking, condition (124) can be replaced by the less restrictive condition (122).

et al. [57, 60] considered the case in which $\Sigma_R = \Sigma_I$ is allowed to have non-zero off-diagonal elements, where the off-diagonal elements represent the correlations between each pair of coils. Holding on assumption (124), this yields $\underline{\mathbf{z}} \sim C\mathcal{N}(\mathbb{E}[\underline{\mathbf{z}}], \Sigma_z)$, with $\Sigma_z = 2\Sigma_R$. In this more general case, the PDF of \mathbf{m}_L^2 cannot be derived, but the mean and variance are given by [19, 57]

$$\mathbb{E}\left[\mathbf{m}_{L}^{2}\right] = a^{2} + 2tr(\Sigma_{R}), \qquad (146)$$

$$Var\left(\mathbf{m}_{L}^{2}\right) = 4\underline{s}^{H}\Sigma_{R}\underline{s} + 4||\Sigma_{R}||_{F}^{2} = 4\underline{s}^{H}\Sigma_{R}\underline{s} + 4tr(\Sigma_{R}^{2}), \quad (147)$$

with $||\cdot||_F$ the Frobenius norm and $tr(\cdot)$ the trace operator. Note that the mean (146) remains unaffected by noise correlation.

Aja-Fernandez et al. [60] show that although the data in this case is not strictly $nc - \chi^2$ distributed, in practical cases this distribution is still a very accurate approximation if so-called *effective* parameters are considered. By using the method of moments, the so-called effective number of coils L_{eff} and the effective noise variance σ_{eff}^2 can be derived [60]:

$$L_{eff} = \frac{a^2 tr(\Sigma_R) + (tr(\Sigma_R))^2}{\underline{s}^H \Sigma_R \underline{s} + ||\Sigma_R||_F^2},$$
(148)

$$\sigma_{eff}^2 = \frac{tr(\Sigma_R)}{L_{eff}}.$$
(149)

Generally, noise correlations will reduce the number of degrees of freedom of the $nc - \chi^2$ distribution and increase the effective variance of the noise. Note that L_{eff} and σ_{eff}^2 depend on the signal and hence on the position within the image. As a result, the statistics of the noise will be spatially variant, and the noise becomes non-stationary [73].
5. Parameter estimation

Now that we have analyzed the statistical distribution(s) of MR images, we will next show how knowledge of this distribution can be used to estimate parameters from these images with optimal accuracy and precision. Furthermore, we will address the question as to what precision may be achieved ultimately from a particular MR image.

Suppose that one wants to estimate a parameter vector $\underline{\theta} = (\theta_1, \theta_2, \dots, \theta_K)^T$ from a set of N data points (i.e., observations) $\mathbf{w}_1, \dots, \mathbf{w}_N$ that have a joint PDF

$$f_{\mathbf{w}_1,\mathbf{w}_2,\cdots,\mathbf{w}_N}(w_1,w_2,\cdots,w_N;\underline{\theta}),\tag{150}$$

which depends on $\underline{\theta}$. The observations may represent pixel values of an MR complex, magnitude or phase image, whereas the parameters $\underline{\theta}$ may, for example, represent the underlying true amplitude and phase values [2, 46], or proton densities and relaxation times [5]. In subsection 5.1, it will be shown how the parameterized PDF (150) can be used to compute the so-called Cramér-Rao lower bound (CRLB), which is a lower bound on the variance of any unbiased estimator of the parameters. Then, in subsection 5.2, it will be shown how from the same PDF the maximum likelihood (ML) estimator, having favorable statistical properties, may be derived.

5.1. The Cramér-Rao lower bound

Obviously, different estimators can be used to estimate $\underline{\theta}$. To assess and compare their performances, quality measures such as accuracy and precision can be used. The accuracy of an estimator is expressed in terms of its bias, which is defined as the deviation of the expected value of the estimator from the true value of the parameter:

$$b(\hat{\underline{\theta}}) = \mathbb{E}[\hat{\underline{\theta}}] - \underline{\theta}.$$
 (151)

The bias represents the systematic error. If the bias of an an estimator is zero, the estimator is called unbiased. The precision of an estimator is defined by its variance, or, in the more general case of vector valued parameters, by its covariance matrix:

$$cov(\underline{\hat{\theta}}) = \mathbb{E}\left[\left(\underline{\hat{\theta}} - \mathbb{E}\left[\underline{\hat{\theta}}\right]\right)\left(\underline{\hat{\theta}} - \mathbb{E}\left[\underline{\hat{\theta}}\right]\right)^{T}\right].$$
 (152)

The diagonal elements of $cov(\hat{\theta})$ represent the variances of the elements of $\hat{\theta}$, whereas the non-diagonal elements represent the covariances between the elements of the estimator. Note, that precision is a measure of the non-systematic error. It concerns the spread of the estimates if the experiment is repeated under the same conditions. Generally, different estimators will have different precisions. However, it can be shown that under general regularity conditions the covariance matrix of any unbiased estimator $\hat{\theta}$ satisfies [74]

$$cov\left(\hat{\underline{\theta}}\right) \ge F^{-1},$$
 (153)

with

$$F = -\mathbb{E}\left[\frac{\partial^2 \ln f_{\mathbf{w}_1, \mathbf{w}_2, \cdots, \mathbf{w}_N}(\mathbf{w}_1, \mathbf{w}_2, \cdots, \mathbf{w}_N; \underline{\theta})}{\partial \underline{\theta} \partial \underline{\theta}^T}\right]$$
(154)

the so-called called *Fisher information matrix*. Inequality (153) expresses that the difference between the left-hand and right-hand member is positive semi-definite. A property of a positive semi-definite matrix is that its diagonal elements cannot be negative. This means that the diagonal elements of $cov\left(\hat{\underline{\theta}}\right)$, that is, the variances of the elements of $\hat{\underline{\theta}}$, are larger than or equal to the corresponding diagonal elements of F^{-1} . Hence, F^{-1} represents a lower bound to the variances of all unbiased $\underline{\hat{\theta}}$. The matrix F^{-1} is called the *Cramér-Rao Lower Bound (CRLB)*.

5.2. Maximum Likelihood estimation

To construct the Maximum Likelihood (ML) estimator of the unknown parameter $\underline{\theta}$ from a set of available observations w_1, \ldots, w_N , we substitute these observations (i.e., numbers) for the corresponding independent variables in equation (150). The expression that results depends only on the unknown parameters $\underline{\theta}$. If we now regard these parameters as variables, the deterministic function

$$L(\underline{\theta}; w_1, w_2, \cdots, w_N) \tag{155}$$

that results is called the *likelihood function*. The Maximum Likelihood estimate $\hat{\underline{\theta}}_{ML}$ of the parameter $\underline{\theta}$ is now defined as the value of $\underline{\theta}$ that maximizes the likelihood function:

$$\left\{\underline{\hat{\theta}}_{ML}\right\} = \arg\max_{\underline{\theta}} L(\underline{\theta}; w_1, w_2, \cdots, w_N)$$
(156)

or, equivalently,

$$\left\{\underline{\hat{\theta}}_{ML}\right\} = \arg\max_{\underline{\theta}} \ln L(\underline{\theta}; w_1, w_2, \cdots, w_N)$$
(157)

in which $\ln L(\cdot)$ is called the *log-likelihood function*. The ML estimator has a number of favorable statistical properties [75]. First, it can be shown that this estimator achieves the CRLB asymptotically, that is, for an infinite number of observations. Therefore, it is asymptotically most precise (or, *asymptotically efficient*). Second, it can be shown that the ML estimator is consistent,

which means that it converges to the true value of the parameter in a statistically well defined way if the number of observations increases. Third, the ML estimator is asymptotically normally distributed, with a mean equal to the true value of the parameter and a covariance matrix equal to the CRLB. If these asymptotic properties also apply to a finite or even small number of observations can often only be assessed by estimating from artificial, simulated observations. Finally, the ML estimator is known to have the invariance property. That is, if $\hat{\underline{\theta}}_{ML}$ is the ML estimator of $\underline{\theta}$, and if $g(\underline{\theta})$ is any function of $\underline{\theta}$, then the ML estimator of $\alpha = g(\underline{\theta})$ is given by $\hat{\alpha} = g(\hat{\underline{\theta}}_{ML})$.

Note that the observations (i.e., pixels) m_1, \ldots, m_N from which the parameter of interest $\underline{\theta}$ is estimated (using the ML estimator) can be selected locally or non-locally. In the first case, the parameter is estimated from pixels in a local neighborhood within which the parameter is assumed to be constant (cfr., [46][66]). In the second case, the pixels for the ML estimation of the true underlying parameter are selected in a non local way based on, for example, the intensity similarity of the pixel neighborhoods. This similarity can be measured using, for example, the Euclidean distance [76] [77], sparseness in a transform domain [78, 79], or, as proposed recently, the Kolmogorov-Smirnov (KS) test [80].

6. Estimation of signal and noise from magnitude images

To illustrate the practical application of the theory summarized in section 5, we will now consider, as an illustrative example, the problem of estimating the underlying true signal amplitude from a single coil magnitude image. For this estimation problem, the CRLB and the ML estimator are derived. For the case of multiple coil images, the CRLB and ML estimator can be derived in a similar way (see, e.g., [67, 81]).

Consider a set of N independent pixel values of a magnitude image taken from a region in which the underlying true signal amplitude a is assumed to be constant. The joint PDF $f_{\mathbf{m}_1,\mathbf{m}_2,\cdots,\mathbf{m}_N}(m_1,m_2,\cdots,m_N)$ of these pixel values (from now on called *observations*) is then given by the product of the marginal PDFs $f_{\mathbf{m}_i}(m_i)$ of the individual observations constituting this set:

$$f_{\mathbf{m}_1,\mathbf{m}_2,\cdots,\mathbf{m}_N}(m_1,m_2,\cdots,m_N;a,\sigma^2) = \prod_{i=1}^N f_{\mathbf{m}_i}(m;a,\sigma^2).$$
(158)

For single-coil images, $f_{\mathbf{m}_i}(m_i; a, \sigma^2)$ is given by the Rician PDF (78). Note that the joint PDF depends on the true signal amplitude a and the noise standard deviation σ , as expressed in the notation used.

6.1. CRLB

The CRLB can be derived from Eqs. (153-154) and (158) and (78). If the noise variance σ^2 is known, the CRLB, is given by [9]:

$$CRLB = \frac{\sigma^2}{N} \left(\eta - \frac{a^2}{\sigma^2} \right)^{-1} \quad , \tag{159}$$

with

$$\eta = \mathbb{E}\left[\frac{\mathbf{m}^2}{\sigma^2} \frac{I_1^2\left(\frac{a\mathbf{m}}{\sigma^2}\right)}{I_0^2\left(\frac{a\mathbf{m}}{\sigma^2}\right)}\right] \quad , \tag{160}$$

The expectation value in (160) can be evaluated numerically. If the noise variance σ^2 is unknown and has to be estimated simultaneously with the signal parameter a, the elements of the Fisher information matrix F with

respect to the parameter vector $\underline{\theta} = (a, \sigma^2)^T$ are given by [2]:

$$F(1,1) = \frac{N}{\sigma^2} \left(\eta - \frac{a^2}{\sigma^2} \right)$$
(161)

$$F(1,2) = F(2,1) = \frac{Na}{\sigma^4} \left(1 + \frac{a^2}{\sigma^2} - \eta \right)$$
(162)

$$F(2,2) = \frac{N}{\sigma^4} \left(1 + \frac{a^2}{\sigma^2} (\eta - 1) - \frac{a^4}{\sigma^4} \right)$$
(163)

where F(i, j) denotes the (i, j)th element of the matrix F and η is given by (160). Finally, the CRLB for unbiased estimation of (a, σ^2) is obtained by simple inversion of the 2 × 2 matrix F.

6.2. Maximum Likelihood estimator

To construct the Maximum Likelihood (ML) estimator of the unknown parameters a and σ^2 from a set of available magnitude observations m_1, \ldots, m_N we substitute these observations (i.e., numbers) for the corresponding independent variables in Equation (158). The thus obtained Likelihood function is then given by

$$L(a, \sigma^{2}; m_{1}, m_{2}, \cdots, m_{N}) = \prod_{i=1}^{N} \frac{m_{i}}{\sigma^{2}} e^{-\frac{a^{2} + m_{i}^{2}}{2\sigma^{2}}} I_{0}\left(\frac{ma}{\sigma^{2}}\right) \epsilon(m)$$
(164)

and the Maximum Likelihood estimates \hat{a}_{ML} and $\hat{\sigma}_{ML}$ of the parameters aand σ are found by maximizing the likelihood function with respect to a and σ [2]:

$$\left\{\hat{a}_{ML}, \hat{\sigma^2}_{ML}\right\} = \arg\max_{a,\sigma^2} \prod_{i=1}^N \frac{m_i}{\sigma^2} e^{-\frac{a^2 + m_i^2}{2\sigma^2}} I_0\left(\frac{ma}{\sigma^2}\right), \quad (165)$$

or, equivalently, since the logarithm is a monotonically increasing function,

$$\left\{ \hat{a}_{ML}, \hat{\sigma}^{2}_{ML} \right\} = \arg \max_{a,\sigma^{2}} \ln L(a,\sigma^{2}; m_{1}, m_{2}, \cdots, m_{N})$$

$$= \arg \max_{a,\sigma^{2}} \ln \prod_{i=1}^{N} \frac{m_{i}}{\sigma^{2}} e^{-\frac{a^{2}+m_{i}^{2}}{2\sigma^{2}}} I_{0}\left(\frac{ma}{\sigma^{2}}\right)$$

$$= \arg \max_{a,\sigma^{2}} \left(\sum_{i=1}^{N} \ln \left(\frac{m_{i}}{\sigma^{2}}\right) - \sum_{i=1}^{N} \frac{m_{i}^{2}+a^{2}}{2\sigma^{2}} + \sum_{i=1}^{N} \ln I_{0}\left(\frac{am_{n}}{\sigma^{2}}\right) \right).$$

$$(166)$$

Note that if the noise variance σ^2 is known, (166) simplifies to

$$\hat{a}_{ML} = \arg \max_{a} \left(\sum_{i=1}^{N} \ln I_0 \left(\frac{am_i}{\sigma^2} \right) - \frac{Na^2}{2\sigma^2} \right).$$
(167)

Yakovleva et al. [82] recently introduced a new technique to calculate the ML estimates of a and σ^2 . This technique effectively reduces the task of solving a system of two nonlinear equations with two unknown variables, to the task of solving just one equation with one unknown variable. Using Yakovleva's technique, finding the ML estimates of both σ^2 and a is therefore not more complicated (in terms of computational cost) than finding the ML estimate of a only (with σ^2 known).

6.3. Discussion

Note that in the more general case, the underlying signal amplitude a can be a parametric function $f(\underline{\theta})$ of an unknown parameter vector $\underline{\theta}$, where typical elements of $\underline{\theta}$ are proton density ρ and relaxation time constants T_1, T_2, T_2^* . In this case, the same theory (described in Section 5) that was used to derive the CRLB and ML estimator of the parameter a, can straightforwardly be applied to derive the CRLB and ML estimator of $\underline{\theta}$ [5]. Furthermore, ML estimates can also be obtained from the complex valued images. It was shown by Sijbers and den Dekker [2] that ML estimation of the underlying true signal amplitude from complex data with equal underlying true phase values is generally better in terms of the mean squared error (MSE) than ML estimation from magnitude data. However, if the phase values vary within the region from which the amplitude is estimated, ML estimation from magnitude data is significantly better in terms of the MSE.

Finally, for a list of series expansions, recursive relations and polynomial approximations of modified Bessel functions that have been proven useful in the numerical calculation of ML estimates from magnitude images, the reader is referred to Appendix C.

7. Discussion and Conclusions

In this review paper, it has been shown that the raw complex MR data points acquired in the spatial frequency domain (i.e., the k-space) are characterized by a joint circularly complex Gaussian distribution, with a diagonal covariance matrix. After taking the inverse DFT, we obtain a complex image in the spatial domain (i.e., the image space). Due to the linearity and the orthogonality of the DFT, the pixels of this so-called reconstructed image are also jointly circularly complex Gaussian distributed with a diagonal covariance matrix. Taking the magnitude and phase, however, are nonlinear operations. Therefore, magnitude and phase images are no longer Gaussian distributed. The PDFs of magnitude and phase images have been described in this paper. In particular, it has been shown that the pixels of magnitude images obtained by single-coil acquisition are Rician distributed, whereas magnitude images acquired using a multiple-coil system (and using the sum-of-squares reconstruction algorithm) are nc- χ distributed, under the assumption that the noise variance at each coil is the same and there are no inter-coil noise correlations. If this assumption is not valid, the noise in the magnitude images becomes spatially non-stationary. In this case, it is no longer possible to derive an exact expression for the distribution of the image pixel values, although, under certain conditions, accurate approximations may still be given.

Furthermore, it has been summarized how knowledge of the distribution of MR images can be used to (i) derive the precision that may be achieved ultimately when estimating parameters from a particular MR image and (ii) to construct the maximum likelihood (ML) estimator, which achieves this precision at least asymptotically.

Finally, we note that data distributions in MR images that were generated using nonlinear reconstruction techniques may be very different from those of conventional Fourier based reconstruction [83–86]. Nonlinear reconstruction techniques have been shown to be successful in reconstructing high-resolution images from subsampled data. Such techniques are becoming increasingly popular, as the demand for shorter scan times without significantly affecting image quality increases. It has been noted [87] that although the convergence and other deterministic properties of nonlinear reconstruction methods are well established, little is known about how noise in the source data influences noise in the final reconstructed image. In [87], the noise distribution from nonlinear reconstructed MR images was determined in an experimental way. Depending on the level of subsampling, the noise distribution was observed to vary from a Rayleigh distribution to a log-normal distribution with increasing level of subsampling. For future research, it would be highly valuable to fully characterize the MR data distribution from solid theoretical derivations.

Appendix A. The complex multivariate Gaussian distribution

The following analysis is to a large extent based on [59]. Let $\underline{\mathbf{z}}_R$ and $\underline{\mathbf{z}}_I$ be the real vectors

$$\underline{\mathbf{z}}_{R} = (\underline{\mathbf{z}}_{R_{1}} \dots \underline{\mathbf{z}}_{R_{L}})^{T}$$
(A.1)

and

$$\underline{\mathbf{z}}_I = (\underline{\mathbf{z}}_{I_1} \dots \underline{\mathbf{z}}_{I_L})^T \tag{A.2}$$

with jointly Gaussian distributed random variables and define

$$\underline{\boldsymbol{t}} = \begin{pmatrix} \underline{\boldsymbol{z}}_R \\ \underline{\boldsymbol{z}}_I \end{pmatrix} \tag{A.3}$$

and

$$\underline{\boldsymbol{z}} = (\boldsymbol{z}_1 \dots \boldsymbol{z}_L)^T, \tag{A.4}$$

with $\boldsymbol{z}_l = \boldsymbol{z}_{R_l} + \imath \boldsymbol{z}_{I_l}$. Next, define

$$\underline{\boldsymbol{w}} = \begin{pmatrix} \underline{\mathbf{z}} \\ \underline{\mathbf{z}}^* \end{pmatrix}. \tag{A.5}$$

The covariance matrix of the complex vector $\underline{\mathbf{z}}$ is defined by its (m, n)th element

$$cov(\mathbf{z}_m, \mathbf{z}_n) = \mathbb{E}[(\mathbf{z}_m - \mathbb{E}[\mathbf{z}_m])(\mathbf{z}_n^* - \mathbb{E}[\mathbf{z}_n^*])].$$
 (A.6)

Therefore,

$$cov(\underline{\mathbf{z}}, \underline{\mathbf{z}}) = \mathbb{E}[(\underline{\mathbf{z}} - \mathbb{E}[\underline{\mathbf{z}}])(\underline{\mathbf{z}} - \mathbb{E}[\underline{\mathbf{z}}])^H].$$
 (A.7)

Similarly,

$$cov(\underline{\mathbf{w}}, \underline{\mathbf{w}}) = \begin{pmatrix} cov(\underline{\mathbf{z}}, \underline{\mathbf{z}}) & cov(\underline{\mathbf{z}}, \underline{\mathbf{z}}^*) \\ cov(\underline{\mathbf{z}}^*, \underline{\mathbf{z}}) & cov(\underline{\mathbf{z}}^*, \underline{\mathbf{z}}^*) \end{pmatrix},$$
(A.8)

with $cov(\underline{\mathbf{z}}^*, \underline{\mathbf{z}}) = cov(\underline{\mathbf{z}}, \underline{\mathbf{z}}^*)^H$, where $cov(\underline{\mathbf{z}}, \underline{\mathbf{z}}^*)$ is known as the *pseudo-covariance* matrix of $\underline{\mathbf{z}}$ [88]. Furthermore, it can be shown that $cov(\mathbf{z}, \mathbf{z})$ and $cov(\mathbf{z}, \mathbf{z}^*)$ can be expressed in terms of the covariance matrices of $\underline{\mathbf{z}}_R$ and $\underline{\mathbf{z}}_I$:

$$cov(\underline{\mathbf{z}}, \underline{\mathbf{z}}^*) = \Sigma_R - \Sigma_I + i(\Sigma_{IR} + \Sigma_{RI}),$$
 (A.9)

$$cov(\underline{\mathbf{z}}, \underline{\mathbf{z}}) = \Sigma_R + \Sigma_I + \imath (\Sigma_{IR} - \Sigma_{RI}).$$
 (A.10)

with

$$\Sigma_R = cov(\underline{\mathbf{z}}_R, \underline{\mathbf{z}}_R), \qquad (A.11)$$

$$\Sigma_I = cov(\underline{\mathbf{z}}_I, \underline{\mathbf{z}}_I), \qquad (A.12)$$

and

$$\Sigma_{IR} = \Sigma_{RI}^T = cov(\underline{\mathbf{z}}_I, \underline{\mathbf{z}}_R).$$
(A.13)

Since the elements of t are jointly Gaussian distributed, it can be shown that the PDF of \underline{z} is given by [59]:

$$\frac{1}{\pi^L \sqrt{\det(\Sigma_w)}} \exp\left\{-\frac{1}{2}(\underline{w} - \mathbb{E}[\underline{\mathbf{w}}])^H \Sigma_w^{-1}(\underline{w} - \mathbb{E}[\underline{\mathbf{w}}])\right\}$$
(A.14)

with $\Sigma_w = cov(\mathbf{w}, \mathbf{w})$. This is usually denoted as $\underline{\mathbf{z}} \sim C\mathcal{N}(\mathbb{E}[\underline{\mathbf{z}}], \Sigma_z, C)$, with $\Sigma_z = cov(\underline{\mathbf{z}}, \underline{\mathbf{z}})$ the covariance matrix of $\underline{\mathbf{z}}$ and $C = cov(\underline{\mathbf{z}}, \underline{\mathbf{z}}^*)$ the pseudocovariance matrix of $\underline{\mathbf{z}}$.

Next, consider the special case that

$$\mathbb{E}[(\mathbf{z}_m - \mathbb{E}[\mathbf{z}_m])(\mathbf{z}_n - \mathbb{E}[\mathbf{z}_n])] = \mathbb{E}[(\mathbf{z}_m - \mathbb{E}[\mathbf{z}_m])^*(\mathbf{z}_n - \mathbb{E}[\mathbf{z}_n])^*] = 0, \quad (A.15)$$

that is, $cov(\underline{\mathbf{z}}, \underline{\mathbf{z}}^*)$ and $cov(\underline{\mathbf{z}}^*, \underline{\mathbf{z}})$ are $L \times L$ null matrices. Then, the complex random variables $\underline{\mathbf{z}}$ are called *circularly complex* and

$$\Sigma_w = \begin{pmatrix} \Sigma_z & O \\ O & \Sigma_z^* \end{pmatrix}, \tag{A.16}$$

where O is the $L \times L$ null matrix. Substituting (A.16) in (A.14) yields the so-called joint circularly complex normal distribution [59], also known as the complex multivariate normal (or Gaussian) PDF [19, 27]:

$$f_{\underline{\mathbf{z}}}(\underline{z}) = \frac{1}{\pi^L \det(\Sigma_z)} \exp\left\{-(\underline{z} - \mathbb{E}[\underline{\mathbf{z}}])^H \Sigma_z^{-1}(\underline{z} - \mathbb{E}[\underline{\mathbf{z}}])\right\}.$$
 (A.17)

This is usually denoted as $\underline{\mathbf{z}} \sim \mathcal{CN}(\mathbb{E}[\underline{\mathbf{z}}], \Sigma_z)$. Note that complex random variables for which condition (A.15) holds (i.e., with a vanishing pseudocovariance matrix) are often called *proper* [88]. Furthermore, note that it follows from (A.9) and (A.10) that condition (A.15) is satisfied if and only if $\Sigma_R = \Sigma_I$ and $\Sigma_{IR} = -\Sigma_{RI}(= -\Sigma_{IR}^T)$, where the skew-symmetry of Σ_{IR} implies that Σ_{IR} has a zero main diagonal, which means that the real and imaginary part of each component \mathbf{z}_k of $\underline{\mathbf{z}}$ are uncorrelated. Note that condition (A.15) also implies that for zero mean $\underline{\mathbf{z}}$ we have $\mathbb{E}[\mathbf{z}_k \mathbf{z}_l] = 0$. The vanishing of $cov(\underline{\mathbf{z}}, \underline{\mathbf{z}}^*)$ and $cov(\underline{\mathbf{z}}^*, \underline{\mathbf{z}})$ does not, however, imply that the real part of \mathbf{z}_k and the imaginary part of \mathbf{z}_l are uncorrelated for $k \neq l$ [88]. Finally, note that if condition (A.15) is satisfied, the covariance matrix Σ_z can be written as

$$\Sigma_z = 2\Sigma_R + 2i\Sigma_{IR}.\tag{A.18}$$

Appendix B. Covariance of two-dimensional DFT

Suppose that the 2D data set in the k-space consists of $M \times M$ complex data points. These data points can be described by a $M \times M$ matrix \boldsymbol{Z} ,

but the data can also be described by a vector of $N = M^2$ elements that is obtained by stacking the columns of \boldsymbol{Z} . Let's define this vector by

$$\underline{\mathbf{z}}(\underline{k}) = \begin{pmatrix} \mathbf{z}(\underline{k}_1) \\ \vdots \\ \mathbf{z}(\underline{k}_N) \end{pmatrix}, \tag{B.1}$$

with

$$\mathbb{E}[\underline{\mathbf{z}}(\underline{k})] = \begin{pmatrix} s(\underline{k}_1) \\ \vdots \\ s(\underline{k}_N) \end{pmatrix}.$$
 (B.2)

Similarly, the 2D data in the image space obtained by applying the 2D inverse discrete Fourier transform is described by the $N \times 1$ vector

$$\underline{\mathbf{z}}(\underline{r}) = \begin{pmatrix} \mathbf{z}(\underline{r}_1) \\ \vdots \\ \mathbf{z}(\underline{r}_N) \end{pmatrix}.$$
(B.3)

Now, it can be shown that

$$\underline{\mathbf{z}}(\underline{r}) = \frac{1}{N} X^H \underline{\mathbf{z}}(\underline{k}), \tag{B.4}$$

where X is an $N \times N$ matrix given by

$$X = A \otimes A, \tag{B.5}$$

where \otimes denotes the Kronecker product and

$$A = \begin{pmatrix} \alpha^{0} & \alpha^{0} & \alpha^{0} & \cdots & \alpha^{0} \\ \alpha^{0} & \alpha^{1} & \alpha^{2} & \cdots & \alpha^{M-1} \\ \alpha^{0} & \alpha^{2} & \alpha^{4} & \cdots & \alpha^{M-1} \\ \vdots & \vdots & \ddots & \vdots \\ \alpha^{0} & \alpha^{(M-1)} & \cdots & \alpha^{(M-1)^{2}} \end{pmatrix},$$
(B.6)

with $\alpha = e^{-i2\pi/M}$. Note that (B.4) is a linear operation. This implies that if $\underline{\mathbf{z}}(\underline{k})$ is Gaussian distributed, its inverse DFT $\underline{\mathbf{z}}(\underline{r})$ will also be Gaussian distributed. Furthermore, it can be shown straightforwardly that the covariance matrix of $\underline{\mathbf{z}}(\underline{r})$ is given by

$$Cov\left(\underline{\mathbf{z}}(\underline{r}), \underline{\mathbf{z}}(\underline{r})\right) = \mathbb{E}\left[\left(\underline{\mathbf{z}}(\underline{r}) - \mathbb{E}[\underline{\mathbf{z}}(\underline{r})]\right)\left(\underline{\mathbf{z}}(\underline{r}) - \mathbb{E}[\underline{\mathbf{z}}(\underline{r})]\right)^{H}\right] = \frac{1}{N^{2}} X^{H} \Sigma_{z} X, \quad (B.7)$$

with

$$\Sigma_z = cov(\underline{\mathbf{z}}(\underline{k}), \underline{\mathbf{z}}(\underline{k})). \tag{B.8}$$

Note that if $\Sigma_z = 2\sigma_K^2 I_N$, with I_N the $N \times N$ diagonal matrix, expression (B.7) simplifies to

$$Cov\left(\underline{\mathbf{z}}(\underline{r}), \underline{\mathbf{z}}(\underline{r})\right) = \frac{2\sigma_K^2}{N^2} X^H X = \frac{2\sigma_K^2}{N} I_N, \tag{B.9}$$

since $X^H X = N I_N$. Finally, if

$$\underline{\mathbf{z}}(\underline{k}) \sim \mathcal{CN}(\underline{s}, 2\sigma_K^2 I_N), \tag{B.10}$$

then

$$\underline{\mathbf{z}}(\underline{r}) \sim \mathcal{CN}(\frac{1}{N}X^{H}\underline{s}, 2\frac{\sigma_{K}^{2}}{N}I_{N}).$$
(B.11)

Appendix C. Modified Bessel functions of the first kind of integer order

The series expansion of the n^{th} order modified Bessel function of the first kind is given by

$$I_{\mathbf{n}}(x) = \left(\frac{x}{2}\right)^n \sum_{m=0}^{\infty} \frac{(x/2)^{2m}}{m!\Gamma(m+n+1)}.$$
 (C.1)

Furthermore, the following recursive relationships hold:

$$I_{n+1}(x) = -\left(\frac{2n}{x}\right)I_n(x) + I_{n-1}(x)$$
(C.2)

$$I_{n-1}(x) + I_{n+1}(x) = 2I'_n(x),$$
 (C.3)

with $I'_n(x) = \frac{d}{dx}I_n(x)$. Moreover,

$$I_{-n}(x) = I_n(x). \tag{C.4}$$

In the region $x \ll n$, $I_n(x)$ becomes, asymptotically, a simple power of its argument [89]

$$I_n(x) \approx \frac{1}{n!} \left(\frac{x}{2}\right)^n, \quad n \ge 0,$$
 (C.5)

whereas in the region $x \gg n$, $I_n(x)$ is well approximated by [89]

$$I_n(x) \approx \frac{1}{\sqrt{2\pi x}} \exp(x).$$
 (C.6)

For $|x| \leq 3.75$, the following polynomial approximations hold [45]

$$\begin{aligned} x^{\frac{1}{2}}e^{-x}I_{0}(x) &= 0.39894228 + 0.0.1328592t^{-1} + 0.00225319t^{-2} \\ &- 0.00157565t^{-3} + 0.00916281t^{-4} - 0.02057706t^{-5} \\ &+ 0.02635537t^{-6} - 0.01647633t^{-7} \\ &+ 0.00392377t^{-8} + \epsilon_{1}, \end{aligned}$$
(C.7)

and

$$x^{\frac{1}{2}}e^{-x}I_{1}(x) = 0.39894228 - 0.03988024t^{-1}$$

-0.00362018t^{-2} + 0.00163801t^{-3}
-0.01031555t^{-4} + 0.02282967t^{-5}
-0.02895312t^{-6} - 0.01787654t^{-7}
-0.00420059t^{-8} + \epsilon_{2}, \qquad (C.8)

with t = x/3.75, $|\epsilon_1| < 1.9 \times 10^{-7}$ and $|\epsilon_2| < 2.2 \times 10^{-7}$.

Practical algorithms for accurate numerical calculation of Bessel functions using the relations described above can be found in [89].

Acknowledgement. This work was supported by the Fund for Scientific Research-Flanders (FWO), and by the Interuniversity Attraction Poles Program (P7/11) initiated by the Belgian Science Policy Office.

References

- Henkelman RM. Measurement of signal intensities in the presence of noise in MR images. Med Phys 1985;12(2):232–3.
- [2] Sijbers J, den Dekker AJ. Maximum likelihood estimation of signal amplitude and noise variance from MR data. Magn Reson Med 2004;51(3):586–94.
- [3] Aja-Fernández S, Alberola-López C, Westin C. Noise and signal estimation in magnitude MRI and Rician distributed images: a LMMSE approach. IEEE T Med Imaging 2008;17(8):1383–98.
- [4] Liu J, Nieminen A, Koenig JL. Calculation of T₁, T₂ and proton spin density in nuclear magnetic resonance imaging. J Magn Reson 1989;85:95–110.
- [5] Sijbers J, den Dekker AJ, Raman E, Van Dyck D. Parameter estimation from magnitude MR images. Int J Imag Syst Tech 1999;10(2):109–14.
- [6] Whittall KP, MacKay AL. Quantitative analysis of NMR data. J Magn Reson Imaging 1989;84:134–52.

- [7] Whittall KP, Bronskill MJ, Henkelman RM. Investigation of analysis techniques for complicated nmr relaxation data. J Magn Reson 1995;84(2):221–34.
- [8] Bonny JM, Zanca M, Boire JY, Veyre A. T₂ maximum likelihood estimation from multiple spin-echo magnitude images. Magn Reson Med 1996;36:287–93.
- [9] Karlsen OT, Verhagen R, Bovée WM. Parameter estimation from Rician-distributed data sets using a maximum likelihood estimator: application to T₁ and perfusion measurements. Magn Reson Med 1999;41(3):614–23.
- [10] Baselice F, Ferraioli G, Pascazio V. Relaxation time estimation from complex magnetic resonance images. Sensors 2010;10:3611–25.
- [11] Zhang H, Ye Q, Zheng J, Schelbert E, Hitchens T, Ho C. Improve myocardial T₁ measurement in rats with a new regression model: Application tot myocardial infarction and beyond. Magn Reson Med 2013;in press. Doi: 10.1002/mrm.24988.
- [12] Andersen JLR. Maximum a posteriori estimation of diffusion tensor parameters using a Rician noise model: Why, how and but. NeuroImage 2008;42(4):1340–56.
- [13] Veraart J, Rajan J, Peeters R, Leemans A, Sunaert S, Sijbers J. Comprehensive framework for accurate diffusion MRI parameter estimation. Magn Reson Med 2012;70(4):972–84.

- [14] Veraart J, Sijbers J, Sunaert S, Leemans A, Jeurissen B. Weighted linear least squares estimation of diffusion MRI parameters: strengths, limitations, and pitfalls. Neuroimage 2013;81:335–46.
- [15] Liang ZP, Lauterbur P. Principles of Magnetic Resonance Imaging A Signal Processing Perspective. IEEE Press Series in Biomedical Engineering; Bellingham, Washington USA: SPIE Optical Engineering Press; 2000. ISBN 0-7803-4723-4.
- [16] Papoulis A. Probability, Random Variables and Stochastic Processes.
 1st ed.; Tokyo, Japan: McGraw-Hill; 1965.
- [17] Lathi B. An introduction to Random Signals and Communication Theory. London: International Textbook Company; 1968.
- [18] Wright G. Magnetic resonance imaging. IEEE Signal Proc Mag 1997;14(1):56–66.
- [19] Kay SM. Fundamentals of statistical signal processing: estimation theory. Upper Saddle River, New Jersey 07458: Prentice-Hall, Inc.; 1993. ISBN 0-13-345711-7.
- [20] Bedrossian E. A product theorem for Hilbert transforms (1963). Proc IEEE 1963;51:868–9.
- [21] Vlaardingerbroek M, Boer J. Magnetic Resonance Imaging: Theory and Practice. Berlin Heidelberg New York: Springer-Verlag; 1996. ISBN 3-540-60080-9.
- [22] Redpath T. Signal-to-noise ratio in MRI. Brit J Radiol 1998;71:704–7.

- [23] Johnson JB. Thermal agitation of electricity in conductors. Phys Rev 1928;32(1):97–109.
- [24] Nyquist H. Thermal agitation of electric charge in conductors. Phys Rev 1928;32:110.
- [25] Connor FR. Noise. 2nd ed.; London: Edward Arnold; 1982.
- [26] Kuperman V. Magnetic Resonance Imaging Physical Principles and Applications. Academic Press; 2000.
- [27] van den Bos A. The multivariate complex normal distribution a generalization. IEEE T Inform Theory 1995;41(2):537–9.
- [28] Henkelman RM. Erratum: Measurement of signal intensities in the presence of noise in MR images. Med Phys 1986;13(4):544.
- [29] Henkelman RM, Bronskill M. Artifacts in magnetic resonance imaging. Rev of Magn Reson Med 1987;2(1):1–126.
- [30] Noll DC, Meyer CH, Pauly JM, Nishimura DG, Macovski A. A homogeneity correction for magnetic resonance imaging with time-varying gradients. IEEE T Med Imaging 1991;10(4):629–37.
- [31] Andersen AH, Kirsch JE. Analysis of noise in phase contrast MR imaging. Med Phys 1996;23(6):857–69.
- [32] Pelc NJ, Bernstein MA, Shimakawa A, Glover GH. Encoding strategies for three-direction phase-contrast MR imaging of flow. J Magn Reson Imaging 1991;1(4):405–13.

- [33] Gatehouse P, Keegan J, Crowe L, Masood S, Mohiaddin R, Kreitner K, et al. Magnetic resonance image restoration. Eur J Radiol 2005;10(5):2172–84.
- [34] Beerbaum P, Körperich H, Gieseke J, Barth P, Peuster M, Meyer H. Blood flow quantification in adults by phase-contrast MRI combined with SENSE: a validation study. J Cardiovasc Magn R 2005;7:361–369.
- [35] Srichai MB, Lim RP, Wong S, Lee VS. Cardiovascular applications of phase-contrast MRI. Am J Roentgenol 2009;192(3):662–75.
- [36] Rauscher A, Sedlacik J, Barth M, Mentzel HJ, Reichenbach J. Magnetic susceptibility-weighted MR phase imaging of the human brain. Am J Neuroradiol 2005;26:736–42.
- [37] Langkammer C, Schweser F, Krebs N, Deistung A, Goessler W, Scheurer E, et al. Quantitative susceptibility mapping (QSM) as a means to measure brain iron? A post mortem validation study. NeuroImage 2012;62(3):1593–9.
- [38] Liu C, Li W, Johnson GA, Wu B. High-field (9.4 T) MRI of brain dysmyelination by quantitative mapping of magnetic susceptibility. NeuroImage 2011;56(3):930–8.
- [39] Schweser F, Sommer K, Deistung A, Reichenbach JR. Quantitative susceptibility mapping for investigating subtle susceptibility variations in the human brain. NeuroImage 2012;62(3):2083–100.
- [40] Bilgic B, Pfefferbaum A, Rohlfing T, Sullivan EV, Adalsteinsson E. MRI

estimates of brain iron concentration in normal aging using quantitative susceptibility mapping. NeuroImage 2012;59(3):2625–35.

- [41] Picinbono B. On circularity. IEEE T Signal Proces 2008;42(12):3473–82.
- [42] Rice SO. Mathematical analysis of random noise. Bell Syst Tech 1944;23:282–332.
- [43] Edelstein WA, Bottomley PA, Pfeifer LM. A signal-to-noise calibration procedure for NMR imaging systems. Med Phys 1984;11(2):180–5.
- [44] Sijbers J, den Dekker AJ, Verhoye M, Van Audekerke J, Van Dyck D. Estimation of noise from magnitude MR images. Magn Reson Imaging 1998;16(1):87–90.
- [45] Abramowitz M, Stegun IA. Handbook of mathematical functions. New York: Dover Publications; 1970.
- [46] Sijbers J, den Dekker AJ, Scheunders P, Van Dyck D. Maximum likelihood estimation of Rician distribution parameters. IEEE Trans Med Imag 1998;17(3):357–61.
- [47] Sijbers J. Estimation of signal and noise from magnitude MR images. Ph.D. thesis; University of Antwerp; 1998.
- [48] Sijbers J, Poot D, den Dekker AJ, Pintjens W. Automatic estimation of the noise variance from the histogram of a magnetic resonance image. Phys Med Biol 2007;52:1335–48.
- [49] Fu H, Kam P. Exact phase noise model and its application to linear minimum variance estimation of frequency and phase of a noisy sinusoid.

In: Proceedings IEEE 19th Int. Symp. Personal, Indoor, Mobile Radio Commun. (PIIMRC). 2008, p. 15.

- [50] O'Donoughue N, Moura J. On the product of independent complex gaussians. IEEE T Signal Proces 2012;60(3):1050–63.
- [51] Gudbjartsson H, Patz S. The Rician distribution of noisy MRI data. Magn Reson Med 1995;34:910–4.
- [52] Bonny JM, Renou JP, Zanca M. Optimal measurement of magnitude and phase from MR data. J Magn Reson 1996;113:136–44.
- [53] Sijbers J, Van der Linden A. Encyclopedia of Optical Engineering; chap. Magnetic Resonance Imaging. Marcel Dekker; 2003, p. 1237–58. ISBN: 0-8247-4258-3.
- [54] Hayes C, Roemer P. Noise correlations in data simultaneously acquired from multiple surface coil arrays. Magn Reson Med 1990;2:181–91.
- [55] Macovski A. Noise in MRI. Magn Reson Med 1996;36:494–7.
- [56] Blaimer M, Breuer F, Mueller M, Heidemann R, Griswold M, Jakob P. SMASH, SENSE, PILS, GRAPPA - How to choose the optimal method. Top Magn Reson Imag 2006;15(4):223–36.
- [57] Aja-Fernández S, Tristán-Vega A, Hoge WS. Statistical noise analysis in GRAPPA using a parametrized noncentral Chi approximation model. Magn Reson Med 2011;65:1195–206.
- [58] Aja-Fernández S, Vegas-Sánchez-Ferrero G, de Luis-García R, Tristán-Vega A. Noise estimation in magnetic resonance SENSE reconstructed

data. In: Proceedings of the 35th Annual International Conference of the IEEE EMBS. Osaka, Japan; 2013,.

- [59] van den Bos A. Parameter Estimation for Scientists and Engineers. New Jersey: John Wiley and Sons; 2007.
- [60] Aja-Fernández S, Tristán-Vega A. Influence of noise correlation in multiple-coil statistical models with sum of squares reconstruction. Magn Reson Med 2012;67:580–5.
- [61] Roemer PB, Edelstein WA, Hayes CE, Souza SP, Mueller OM. The NMR phased array. Magn Reson Med 1990;16:192–225.
- [62] Miller K. Multivariate Distributions. Augmented ed.; Huntington, New York: Robert E. Krieger Publishing Company; 1975.
- [63] Constantinides CD, Atalar E, McVeigh ER. Signal-to-noise measurements in magnitude images from NMR phased arrays. Magn Reson Med 1997;38:852–7.
- [64] Mood AM, Graybill FA, Boes DC. Introduction to the Theory of Statistics. 3rd ed.; Tokyo: McGraw-Hill; 1974.
- [65] Dietrich O, Raya J, Reeder S, Ingrisch M, Reiser M, Schoenberg S. Influence of multichannel combination, parallel imaging and other reconstruction techniques on MRI noise characteristics. Magn Reson Imaging 2008;26:754–62.
- [66] den Dekker AJ, Sijbers J. Advanced Image Processing in Magnetic Resonance Imaging; vol. 27 of Signal Processing and Communications;

chap. 4: Estimation of signal and noise from MR data. CRC press; 2005,p. 85–143.

- [67] Rajan J. Estimation and removal of noise from single and multiple coil magnetic resonance images. 2012. PhD Thesis, University of Antwerp.
- [68] Kay SM. Fundamentals of Statistical Signal Processing, Volume II Detection Theory. Upper Saddle River, New Jersey: Prentice Hall PTR; 1998.
- [69] Simon M. Probability distributions involving Gaussian random variables - A handbook for engineers and scientists. Norwell, Massachusetts, U.S.A.: Kluwer Academic Publishers; 2002.
- [70] den Dekker AJ, Sijbers J, Verhoye M, Van Dyck D. Maximum likelihood signal estimation in phase contrast magnitude MR images. In: Proceedings of SPIE: Medical Imaging; vol. 3338. San Diego, CA, USA; 1998, p. 408–15.
- [71] Harpen M. Noise correlations exist for independent RF coils. Magn Reson Med 1992;23:394 –7.
- [72] Pruessmann K, Weiger M, Scheidegger M, Boesiger P. SENSE: Sensitivity encoding for fast MRI. Magn Reson Med 1999;42:952–62.
- [73] Aja-Fernández S, Brion V, Tristán-Vega A. Effective noise estimation and filtering from correlated multiple-coil MR data. Magn Reson Imaging 2013;31:272–85.

- [74] van den Bos A. Handbook of Measurement Science; vol. 1; chap. 8:
 Parameter Estimation. Wiley, Chichester, England: Edited by P. H. Sydenham; 1982, p. 331–77.
- [75] Cramér H. Mathematical Methods of Statistics. Princeton, NJ: University Press; 1946.
- [76] He L, Greenshields IR. A nonlocal maximum likelihood estimation method for rician noise reduction in MR images. IEEE T Med Imaging 2009;28(2):165 –72.
- [77] Rajan J, Van Audekerke J, Verhoye M, Sijbers J. Nonlocal maximum likelihood estimation method for denoising multiple-coil magnetic resonance images. Magn Reson Imaging 2012;30:1512–8.
- [78] Manjon JV, Coupe P, Buades A, Collins DL, Robles M. New methods for MRI denoising based on sparseness and self-similarity. Medical Image Analysis 2012;16(1):18–27.
- [79] Maggioni M, Katkovnik V, Egiazarian K, Foi A. A nonlocal transformdomain filter for volumetric data denoising and reconstruction. IEEE T Imag Proc 2013;22(1):119–33.
- [80] Rajan J, den Dekker AJ, Sijbers J. A new non local maximum likelihood estimation method for Rician noise reduction in magnetic resonance images using the Kolmogorov-Smirnov test. Signal Process 2014;doi:http://dx.doi.org/10.1016/j.sigpro.2013.12.018.
- [81] Beltrachini L, von Ellenrieder N, Muravchik C. Error bounds in diffusion

tensor estimation using multiple-coil acquisition systems. Magn Reson Imaging 2013;31:1372–83.

- [82] Yakovleva T, Kulberg N. Noise and signal estimation in mri: twoparametric analysis of rice-distributed data by means of the maximum likelihood approach. AJTAS 2013;2(3):67–79.
- [83] Liang ZP, Boada F, Constable R, Haacke E, Lauterbur P, Smith M. Constrained reconstruction methods in MR imaging. Rev Magn Reson Med 1992;4(2):67–185.
- [84] Lustig M, Donoho D, Pauly J. Sparse MRI: the application of compressed sensing for rapid MR imaging. Magn Reson Med 2007;58(6):1182–95.
- [85] Lustig M, Donoho D, Santos J, Pauly J. Compressed sensing MRI. IEEE Signal Proc Mag 2008;72:1014–32.
- [86] Uecker M, Hohage T, Block K, Frahm J. Image reconstruction by regularized nonlinear inversion—joint estimation of coil sensitivities and image content. Magn Reson Med 2008;60(3):674–82.
- [87] Sabati M, Peng H, Lauzon ML, Frayne R. A statistical method for characterizing the noise in nonlinearly reconstructed images from undersampled MR data: The POCS example. Magn Reson Imaging 2013;31:1587– 98.
- [88] Neeser F, Massey J. Proper complex random processes with applications to information theory. IEEE T Inform Theory 1993;39(4):1293–302.

[89] Press WH, Vettering WT, Teukolsky SA, Flannery BP. Numerical Recipes in C. Cambridge: University Press; 1994. Chap. 10.